CAPITAL UNIVERSITY OF SCIENCE AND TECHNOLOGY, ISLAMABAD



Properties of Concrete Having Used Petrol-Engine Oil and Varying Lengths of Banana Fibers

by

Muhammad Jawad Khan

A thesis submitted in partial fulfillment for the degree of Master of Science

in the Faculty of Engineering Department of Civil Engineering

2023

Copyright \bigodot 2023 by Muhammad Jawad Khan

All rights reserved. No part of this thesis may be reproduced, distributed, or transmitted in any form or by any means, including photocopying, recording, or other electronic or mechanical methods, by any information storage and retrieval system without the prior written permission of the author. I want to dedicate this work to my family, who has helped me throughout my education. This is also a homage to our finest professors who have inspired us to face the challenges of presence with creativity, and courage and who have made us what we are today.



Properties of Concrete Having Used Petrol-Engine Oil and Varying Lengths of Banana Fibers

by

Muhammad Jawad Khan Registration No: (MCE203034)

THESIS EXAMINING COMMITTEE

S. No.	Examiner	Name	Organization
(a)	External Examiner	Dr. Zeeshan Alam	IIUI, Islamabad
(b)	Internal Examiner	Dr. Nafeesa Shaheen	CUST, Islamabad
(c)	Supervisor	Dr. Majid Ali	CUST, Islamabad

Dr. Majid Ali Thesis Supervisor April, 2023

Dr. Ishtiaq Hassan Head Dept. of Civil Engineering April, 2023 Dr. Imtiaz Ahmad Taj Dean Faculty of Engineering April, 2023

Author's Declaration

I, Muhammad Jawad Khan hereby state that my MS thesis titled "Properties of Concrete Having Used Petrol-Engine Oil and Varying Lengths of Banana Fibers" is my own work and has not been submitted previously by me for taking any degree from Capital University of Science and Technology, Islamabad or anywhere else in the country/abroad.

At any time if my statement is found to be incorrect even after my graduation, the University has the right to withdraw my MS Degree.

Muhammad Jawad Khan

Plagiarism Undertaking

I solemnly declare that the research work presented in this thesis titled "**Properties** of Concrete Having Used Petrol-Engine Oil and Varying Lengths of Banana Fibers" is solely my research work with no significant contribution from any other person. Small contribution/help wherever taken has been duly acknowledged and that complete thesis has been written by me.

I understand the zero tolerance policy of the HEC and Capital University of Science and Technology towards plagiarism. Therefore, I (as an author of the above titled thesis) declare that no portion of my thesis has been plagiarized and any material used as reference is properly referred/cited.

I, undertake that, if I am found guilty of any formal plagiarism in the above titled thesis even after award of MS Degree, the University reserves the right to withdraw/revoke my MS degree and that HEC and the University have the right to publish my name on the HEC/University website on which names of students are placed who submitted plagiarized work.

Muhammad Jawad Khan

List of Publications

It is certified that following publication(s) have been made out of the research work that has been carried out for this thesis:-

Conference Proceedings

Muhammad Jawad Khan

Acknowledgement

In the Name of **Allah**, The Most Gracious, The Most Merciful. Praise be to God, the Cherisher and Sustainer of the worlds. All thanks to Almighty **Allah**, The Lord of all that exist, who bestowed me with His greatest blessing i.e. knowledge and Wisdom to accomplish my task successfully. Thousands of salutations and benedictions to the **Holy Prophet Hazrat Muhammad (PBUH)** the chosenthrough by whom grace the sacred Quran was descended from the Most High.

I am very thankful to Engr. Prof. Dr. Majid Ali, a great teacher, supervisor and trainer who made a difference in all aspects of my life. I am indebted to Engr. Prof. Dr. Majid Ali for his valuable guidance, encouragement and dedicated support that enabled me to complete my MS Degree Program.

I would like to thanks **Engr. Muhammad Adnan** for his helping hand. I want to express my heartiest regards to my parents, my family specially my cousin **Engr. Muhammad Waqas** who always supported me morally, spiritually prayed for my success.

Muhammad Jawad Khan

Abstract

Concrete is a widely used construction material due to its superior properties, which make it more favorable than any other construction material. However, concrete has some serious drawbacks that are caused by its brittle nature, including vulnerability to cracks, brittle behavior, spalling, and more linear shrinkage. Even though concrete has high compressive strength, it is weak in tension, which leads to cracking and reduces the structure's durability and service life. To address these issues, there are different approaches that can be used to enhance the splitting-tensile and flexural properties of concrete. For example, natural fibers and admixtures can be used to improve the properties of concrete. Additionally, there is a growing interest in using residual waste as a raw material for the manufacturing of sustainable materials. However, some residual wastes can cause severe damage to the environment, such as used petrol-engine oil. On the other hand, agricultural waste, such as banana fibers, which are abundantly available in developing countries, can be used as dispersed reinforcement to improve the properties of concrete. Overall, the use of natural fibers and residual waste as a raw material can help in the development of environmentally friendly materials. Therefore, banana fibers can be used to enhance the properties of concrete, and used petrol-engine oil can be used as a chemical during the manufacturing of concrete.

The purpose of this research project is to examine the use of chemical admixtures and natural fibres in concrete to improve its performance and qualities. Specifically, banana fibers are used as a reinforcement, and used petrol-engine oil is added as a chemical admixture. The study involves preparing four cylinders and two beamlets for plain concrete (PC) and used petrol engine oil banana fiber reinforced concrete (UBFRC). The BF is added to the mixture in fixed proportion of 3.0%, and the length is varied (3+5 cm, 3+7 cm, 5+7 cm, 3+5+7 cm) in combination with the UPEO. The mix design 1:2:4 (cement: sand: aggregates) with a water cement ratio (w/c) of 0.5 is used for preparing PC and used petrol-engine oil plain concrete (UPC). The mix 1:2:4 with a w/c of 0.6 is used for preparing fiber reinforced concrete (FRC). For UBFRC, a fixed amount of UPEO is used, and BF is used with a fixed proportion of 3.0% and varying length. Various tests are performed to determine the influence of BF and UPEO on the properties of concrete, including slump, dynamic, mechanical, water absorption, linear shrinkage, and mass-loss tests. The average of two specimens of each type of concrete and test is taken to evaluate every type of the observed property. In summary, this research work explores the use of banana fibers and used petrol-engine oil as a reinforcement and chemical admixture, respectively, to enhance the properties of concrete. The study involved preparing various types of concrete and performing different tests to evaluate the properties of each type.

From obtained results of tests, findings that were collected. The findings of the slump test demonstrate that UPEO-only concrete has a higher slump value than other types of concrete. With an increase in banana fibre content, FRC's workability is seen to decline. Better performance against dynamic loading has been demonstrated by the UBFRC with fibres that are (3+5cm) in length. According to the findings, UBFRC has greater tensile and flexural strengths than other materials when compared to those that include (5+7cm) and (3+5+7cm) amounts of BF, respectively. Additionally, as the length of BF in the concrete increases, the compression toughness index and splitting tensile toughness index both dramatically improve. When specimens' compression strength increases, a drop is observed. With an increase in fibre content, specimens' compression strength begins to decline. Water absorption and linear shrinkage are both increased by adding more fibre to concrete. This leads to the conclusion that (5+7cm) is the ideal fibre content for concrete tension members since it performs better under tensile, flexural, and dynamic loadings. UBFRCs made with a (3+7cm) fibre composition have performed better than other forms of UBFRCs for compression members.

Keywords: Banana Fibers, Dynamic Properties, Fiber Reinforced Concrete, Mechanical Properties, Used Petrol-Engine Oil.

Contents

A	uthor	r's Declaration	iv
Pl	agiar	rism Undertaking	\mathbf{v}
Li	st of	Publications	vi
A	cknov	wledgement	vii
Al	bstra	\mathbf{ct}	viii
Li	st of	Figures	xii
Li	st of	Tables	xiii
Al	bbrev	viations and Symbols	xiv
1	 Intr 1.1 1.2 1.3 1.4 1.5 1.6 1.7 	oductionBackgroundResearch Motivation and Problem Statement1.2.1Research QuestionsOverall Objective of the Research Program and Specific Aim of thisMS ThesisScope of Work and Study Limitations1.4.1Rationale Behind Variable SelectionNovelty of Work, Research Significance and Practical ImplementationBrief MethodologyThesis Outline	1 2 3 4 4 5 1 5 6 8
2	Lite 2.1 2.2 2.3 2.4	rature ReviewBackground	10 10 10 11 14 17 18 19

		 2.4.2 Strength Properties	20 20 21
	2.5	Summary	21
3	Exp	perimental Scheme	23
	3.1	Background	23
	3.2	Raw Constituents	24
	3.3	Mix Design, Concrete Preparation, Fresh Pr- operties, Casting of	
		Specimens and Density Determination of Hard Concrete	25
	3.4	Testing Methodology	30
		3.4.1 Dynamic Tests	30
		3.4.2 Miscellaneous Tests (Water Absorption, Linear Shri- nkage,	91
		and Mass Loss)	31 94
		3.4.3 Mechanical lests	34 25
	25	3.4.4 Visual inspection of Crack Surface	30 26
	5.0	Summary	50
4	Res	ults and Analysis	37
	4.1	Background	37
	4.2	Dynamic Properties	37
	4.3	Miscellaneous Properties	41
		4.3.1 Water Absorption,	41
		4.3.2 Mass Loss	42
		4.3.3 Linear Shrinkage	42
	4.4	Mechanical Properties	43
		4.4.1 Compressive Properties	43
		4.4.2 Splitting Tensile Properties	45
		4.4.3 Flexural Properties	47
	4.5	Fiber Breakage and Pullout	50
	4.6	Summary	52
5	Gui	delines for Practical Implementation	53
	5.1	Background	53
	5.2	Optimization of Banana Fibers with Waste Petrol-Engine Oil	54
	5.3	Application of this Research in Real Life	57
	5.4	Summary	58
C	C	alusions and Decommondations	50
U	$\bigcirc OI$	Conclusions	59 50
	0.1	Future Works	99 60
	0.2		00
Bi	bliog	graphy	61

List of Figures

2.1	Effects of Addition of Waste Engine Oil (UEO) on Concrete's Properties [36]	13
2.2	Schematic Diagram of Fibre Bridging Effect [57]	15
3.1	a) Measuring the Value of Slump of UBFRC, b) Combined effect of UPEO and BF on the relation between slump of fresh concrete and density of hard concrete	28
3.2	Tests Setup: a) Dynamic Test, and b) Mechanical Properties	30
0.2	resus setup, a) Dynamic resu, and b) Meenamear Properties	00
4.1	Compression Behavior ; a) Typical Tested Compression Specimens and b) Stress Strain Curves	44
4.2	Tension Behavior ; a) Typical Tested Tension Specimens and b)	
	Load Deformation Curves	46
4.3	Flexural Behavior ; a) Typical Tested Flexural Specimens and b)	
	Load Deflection Curves	48

List of Tables

3.1	Mix Design, Specimen Labeling, Slump of Fresh Concrete, and Den- sity of Hard Concrete	29
3.2	Standards Testing and Studied Parameters	$\frac{20}{32}$
4.1	Dynamic Properties of PC, UPC, and UBFRCs	40
4.2	Water Absorption, Linear Shrinkage, and Mass Loss of PC, UPC, and UBFRCs	42
4.3	Compressive, Splitting-Tensile, and Flexural Properties of PC, UPC, and all UBFRCs	49
5.1	Optimization of Banana Fiber Content in UBFRC	56

Abbreviations and Symbols

\mathbf{BF}	Banana Fibers
CE1	Compressive Pre-Crack Energy Absorption
CE2	Compressive Post-Crack Energy Absorption
\mathbf{CS}	Compressive Strength
CTI	Compressive Toughness Index
Ed	Dynamic Modulus of Elasticity
\mathbf{FC}	Fiber Content
FE1	Flexural Pre-Crack Energy Absorption
FE2	Flexural Post-Crack Energy Absorption
FRC	Fiber Reinforced Concrete
\mathbf{FS}	Flexural Strength
\mathbf{FTI}	Flexural Toughness Index
MOE	Modulus of Elasticity
\mathbf{PC}	Plain Concrete
Rd	Dynamic Modulus of Rigidity
\mathbf{RFL}	Response Frequencies Lateral
\mathbf{RFR}	Response Frequencies Rotational
\mathbf{RFT}	Response Frequencies Transverse
SE1	Splitting-Tensile Pre-Crack Energy Absorption
SE2	Splitting-Tensile Post-Crack Energy Absorption
\mathbf{SP}	Superplasticizer
\mathbf{SS}	Splitting-Tensile Strength
STI	Splitting-Tensile Toughness Index
\mathbf{STM}	Servo-Hydraulic Testing Machine

UBFRC	Used Petrol-Engine Oil Banana Fiber reinforced concrete
UEO	Used Engine Oil
UPC	Use Petrol-Engine Oil Plain Concrete
UPEO	Used Petrol-Engine Oil
UVBFRC	Used Petrol-Engine Oil and varying length Banana Fiber
ζ	Damping Ratio
Δ	Deflection
δ	Deformation

XV

Chapter 1

Introduction

1.1 Background

In the present era, concrete is the most widely used material in construction. However, it has several limitations such as low tensile strength, vulnerability to cracking, and environmental issues. To overcome these limitations, fibers and admixtures are added to concrete to improve its performance and develop sustainable materials. Fiber reinforced concrete (FRC) is a promising material in civil engineering due to its advantages such as toughness, tensile strength, durability, and energy absorption [1-3]. Because of this fact, cannot be relied on concrete. Also, vulnerability to cracking, loading, and environmental issues are the most promising factors which are causing reduction in functionality and serviceability of concrete[4, 5]. It is required to add fibers and admixtures in concrete to meet with high performance, attaining certain properties, and developing a sustainable material [6].

Admixtures are used for different purposes such as enhancing early strength, accelerating or retarding setting times of concrete or to achieve specific property of the concrete such as to reduce the content of cement while having no effect on the physical properties of the concrete. In past few decades, interest had increased in attaining high performance and environmental friendly materials in civil engineering application. One of these materials, fibre reinforced concrete (FRC),

 $\mathbf{2}$

has gained popularity in civil engineering because to its features, which include toughness, tensile strength, durability and energy absorption [6]. FRC is concrete having dispersed short discrete fibers, this makes FRC to be studied carefully taking in consideration functionality of fiber within the mix. Many researchers attempted to improve the governing properties of concrete for the production of the high-performance concrete. Used petrol-engine oil (UPEO) has been used as a chemical admixture by many researchers for reducing cement content in concrete or as an admixture. Concrete emits carbon di oxide approximately equal to clinker for production. The reduction of cement content in concrete results in reduction in emission of CO2 during the sintering process and making the composite less adverse to the environment. Addition of UPEO reduces 9.4% cement content in concrete with comparable properties of plain concrete.

The addition of fibers to concrete can greatly enhance its mechanical properties such as flexural, tensile strength, resistance against spalling, cracking, and fatigue. Concrete composed of fibers can improve the performance of the concrete for specific applications, including its resistance to impact loading, flexural strength, and splitting tensile strength. In addition, the tensile strength provided by the fibers also helps to resist the production of cracks in the concrete. By reducing cement content and adding used petrol-engine oil (UPEO) as a chemical admixture, the production of concrete can be made more economical. These approaches can effectively enhance numerous mechanical properties of cementitious composites.

1.2 Research Motivation and Problem Statement

Concrete is the primary construction material in this modern age. Although it has many flaws, it cannot be replaced with any other material. Concrete is a such material which inherent certain issues like appearing of cracks in structural members even from day one. These cracks can be found on different structural members. These cracks affect the service life of structure and structural members. If such cracks can be delayed than service life can also be extended. That's why, what can we do is to have sustainable concrete preferably no or minimum cracks at the start. As, the main concern of this research study is to mitigate or reduce the impact of these shortcomings. The failure of concrete structures can result in the loss of human lives and loss of money. That's why it is required to avoid the failure of the concrete in different scenarios of applications and loadings. The use of natural fibers has been reported to enhance the properties of concrete to avoid its failure. There is need to adopt modern methods in the construction industry. Also, it is needed to aware people regarding high-performance and sustainable material. There are also research studies available on improving concrete properties by the addition of the admixture. But there is a need to use the other hazardous/toxic waste materials (like UPEO) to avoid environmental pollution and to take a step towards the development of sustainable and cleaner production. Moreover, this study will help researchers to provide guidelines and a way of thinking to utilize the wastes and environmental pollution causing materials in construction materials instead of dumping of these materials. The dumping of these materials is a time taking and can be costly. Thus, the problem statement is as follow:

In the building construction, use of concrete is becoming inevitable day by day. The concrete is weak in tension, less resistive against lateral loading, and quasi brittle material. The change in temperature cause volumetric change in concrete. This change is the key to production of drying shrinkage cracks. These flaws of concrete are needed to overcome and to improve the properties of concrete. On one side, natural fibers (i.e. banana fibers) can be used to enhance toughness of concrete. And on the other hand, waste materials (like UPEO) can be utilized. There is a need to explore the behavior of concrete having varying length of BF and fixed amount of UPEO.

1.2.1 Research Questions

- What are the combined effects of banana fibers with varying length and used petrol-engine oil on dynamic properties of concrete?
- What are the combined effects of banana fibers with varying length and used petrol-engine oil on mechanical properties of concrete?

- How much water absorption, mass loss and linear shrinkage of concrete be controlled with used petrol-engine oil and varying length banana fiber?
- How much would be the fiber breakage and fiber pullout with the use of UPEO in BFRC?
- How concrete made of used petrol-engine oil and banana fiber with varying length can be used for the specific real-life applications?

1.3 Overall Objective of the Research Program and Specific Aim of this MS Thesis

The overall objective of the research program is to precisely take a step toward development of the high-performance concrete with the help of waste materials for building construction and civil engineering applications. As concrete has several flaws, and some of these flaws affect the performance of structure and decreases the durability of concrete. There is a need to use the waste materials (instead of dumping them) in construction materials because it is reported to have good potential for bringing better impact in cementitious composite. By incorporating waste materials in the construction industry, we can not only reduce the amount of waste going to landfills but also contribute to the development of more sustainable and eco-friendly construction practices.

The specific aim of this research work is to study the combined effect of varying lengths banana fibers (VLBF) and used petrol-engine oil (UPEO) on properties of concrete.

1.4 Scope of Work and Study Limitations

The workability, mechanical properties, dynamic properties and absorption properties of concrete are investigated by taking two specimens for each property of used petrol-engine oil and varying length banana fiber reinforced concrete (UVBFRC) and used petrol-engine oil plain concrete (UPC). The average of two specimens is taken according to acceptance and recommendation of ACI 311.6-18 standards. Dynamic properties are studied before investigating the mechanical and absorption properties. After appearance of first crack on specimen, it is considered as failure after load application. Other miscellaneous properties like linear shrinkage and mass loss are also examined in this study.

The study is purely limited to mechanical, dynamic and absorption properties of UVBFRC specimens. In this scope of study, durability of UVBFRC is not included. Influence on performance and resistance against the impact loading is not considered in this study. The fibers are used with varying length and fix content for different UVBFRCs with fix amount of the UPEO.

1.4.1 Rationale Behind Variable Selection

Fiber's type selected on the superiority of physical properties comparison to other [7]. Banana fibers has high tensile strength among the natural fibers. And also, these banana fibers are likely to be compatible to use with UPEO [8]. The many other natural fibers can easily get damaged by acidic property of UPEO [9]. Different length and size will help in better mixing to achieve good improved properties.

1.5 Novelty of Work, Research Significance and Practical Implementation

In an experimental work, it was revealed that resistance against impact loading significantly improved by the addition of the natural fibers in concrete [10]. The mechanical properties of concrete were observed to be improved by the addition of natural fibers [11]. Previous conducted studies show that properties of concrete and performance of structure were improved with different types of natural fibers and admixtures. To the best of author's knowledge, no research has been conducted on combined effect of used petrol-engine oil and banana fibers on production and properties of concrete. Thus, the current study is aimed to study the basic mechanical, dynamic and absorption properties of UBFRC using used engine oil and banana fibers. This material is resulted in production of improved properties of concrete for the use in civil engineering and construction industry.

There are several flaws of concrete like cracking, spalling, weak in tension, etc. So, there is a need to mitigate these flaws of concrete. The addition of banana fiber and used petrol-engine (UPEO) in concrete enhances resistance against these flaws of concrete. The addition of fibers in concrete resulted in improved durability and enhanced resistance against production and progression of cracks [12]. Fiber reinforced concrete have shown the improved properties in comparison with respect to the properties of the PC. According to a research, fiber reinforced concrete (FRC) beams along with fiber reinforced polymer bars as reinforcement has shown better performance [13]. In Previous studies, single fiber or combination of two fibers was used in concrete for improving its properties. The very limited studies are available in which artificial fibers were used along with an admixture. Therefore, utilization of natural fibers and used petrol engine oil are way better to be used for improving the properties of concrete, as it cleaning the environment as well by using the UPEO. Banana fibers has the high tensile strength in comparison with the other natural fibers [14]. Hence, there is need to investigate its effect on different properties of the concrete.

The concrete having UPEO can directly be practically implemented at locations where it does not have direct contact with environment. For example, this can be used as lean concrete beneath the foundation. Also, lean concrete is covered within the soil and it has no contact with the air and protected from the chemical attacks of environment. However, it seems to have potential for structural applications. If used with great care.

1.6 Brief Methodology

In this experimental study, the basic mechanical, dynamic and absorption properties of Plain Concrete (PC) Used petrol-engine oil Plain Concrete (UPC), and

used petrol-engine oil Plain Concrete banana fibers reinforced concrete UBFRC are determined in laboratory. All UBFRCs are prepared with varying length of banana fibers having fixed content 3%. Fixed amount of used petrol-engine oil plain concrete is used in manufacturing of UPC and all types of UBFRCs. Most conventionally used mix design 1:2:4 is used in manufacturing of PC, UPC, and all UBFRCs. For PC and UPC, 0.5 water-cement ratio (w/c) is used whereas 0.6% w/c is used in making of all types of UBFRCs. The w/c is enhanced due to high water absorption of banana fiber as reported by [15]. The value of w/cis restricted to 0.6 because of addition of UPEO which is, also, a liquid. Other wise, 0.7 w/c ratio had been used due the water absorption of natural fibers in cement composites [2]. The slump cone test method is adopted to evaluate the workability of PC, UPC, and UBFRCs. All specimens are cast and tested as per the ASTM standards. After performing the slump test, total 36 number of specimens are cast of PC, UPC, and UBFRCs. Total four cylinders and 2 beamlets are casted of each type of prepared concrete. Two cylinders of each type are used in investigating the compression properties and rest of two are used in determining the splitting-tensile properties of concrete. Flexural properties of each specimen are determined. But here in case of flexural test only beamlets are used. Before the performance of mechanical testing, the dynamic properties are investigated with the help of accelerometer and a hammer. Response-frequency longitudinal (RFL), response-frequency transvers/lateral (RFT), and response-frequency rotational/torsional (RFR) are noted by attaching the accelerometer to the specimen and applying a stroke of hammer for according specific setup of determination of desired frequencies. These frequencies are than used to evaluate the dynamic properties of the all types of manufactured concrete. The fractured surfaces of broken specimens are closely examined as well to check the mixing of fibers within the concrete, bonding of fibers with surrounding cementitious matrix, fiber pullout, and fiber breakage etc. All specimens are cast and tested as per the ASTM standards. Non-destructive testing are performed before destructive testing. After performing the slump test, total 36 number of specimens are cast of PC, UPC, and UBFRCs. Total four cylinders and 2 beamlets are casted of each type of prepared concrete. Two cylinders of each type are used in investigating the compression properties and rest of two are used in determining the splitting-tensile properties of concrete. Flexural properties of each specimen are determined. But here in case of flexural test only beamlets are used. Before the performance of mechanical testing, the dynamic properties are investigated with the help of accelerometer and a hammer. Response-frequency longitudinal (RFL), response-frequency transvers/lateral (RFT), and response-frequency rota- tional/torsional (RFR) are noted by attaching the accelerometer to the specimen and applying a stroke of hammer for according specific setup of determination of desired frequencies. These frequencies are than used to evaluate the dynamic prop- erties of the all types of manufactured concrete (i.e cylinders and beamlets) The fractured surfaces of broken specimens are closely examined as well to check the mixing of fibers within the concrete, bonding of fibers with surrounding cementitious matrix, fiber pullout, and fiber breakage etc.

1.7 Thesis Outline

The following is a description of the chapters in this thesis:

Chapter 1 introduces the thesis, including the background, research motivation, Problem statement, general and detailed research objectives, work scope, study limitations, methodology, and thesis outline.

Chapter 2 presents a literature review on various topics related to the study, including used petrol-engine oil, banana fibers, recoverable flaws with waste usage, governing values towards improvement in concrete's properties, durability considerations, and design considerations. The chapter concludes with a summary.

Chapter 3 describes the experimental scheme, raw constituents, mix design casting of specimen, testing, and a summary of the chapter.

Chapter 4 presents the detail results obtained from testing and analysis, including the background, dynamic properties and mechanical properties of the mixes (PC, UPC, and UBFRCs), miscellaneous properties, fractured surfaces of tested specimens, and a summary of the chapter. Chapter 6 is the conclusion and future recommendation.

At the end, references are provided.

Chapter 2

Literature Review

2.1 Background

Fiber has been used since ancient times to increase mechanical strength parameters and the performance of composites. Natural fibers are favored since they are economical and eco-friendly. The amount of agricultural waste produced annually is measured in the millions of tons. It is need of the time for effective use of these wastes. It is a proven fact that fiber reinforced concrete performs better than conventional concrete. In order to take a step towards sustainable development, it is necessary to investigate the potential of agricultural wastes (natural fiber) as a sustainable construction material.

2.2 Use of Wastes as Construction Materials

With regards to environmental preservation, the construction industry's interest in using waste products as raw materials has been expanding over time. Numerous studies have examined the effects of various waste products on the characteristics of concrete. A number of researchers have used recycled aggregates in concrete to improve the development of sustainable construction materials and cleaner production [16-19]. Various researchers have used glass fiber plastic waste and ceramic waste in concrete [20-23]. Due to their usage as construction materials, agricultural wastes have a significant economic impact all over the world. Natural fibers have helped to build sustainable and ecologically friendly materials by reducing the negative effects of other common hazardous materials. They are used in many different types of composites. Environmentally hazardous materials include the oil used by various sorts of engines and machinery. In order to produce concrete sustainably and to dispose of used engine oil economically, these waste lubricants can be utilized [24]. A research study was conducted by Hamad et al. to investigate the fresh and hardened properties of concrete having used engine oil [25].

Waste engine oil do not have any significant negative impact on the structural components, according to research on the additive effects of used/waste engine oil on the behavior of reinforced structural elements [26]. Waste engine oil cannot be kept and reused in an efficient manner since there is so much of it accessible. It is important to prevent used engine oil from entering the runoff water and combining with it. In the end, it might contaminate the marine and river ecosystems and damage aquatic life. A significant portion of valuable land is covered by the disposal of this trash, which is also harmful to human health [27]. The agricultural waste can also be burned instead of dumped. When a lot of agricultural waste is burned, a lot of heat is released into the atmosphere, affecting the environment and hastening the effects of global warming.

2.2.1 Used Petrol-Engine Oil

Vehicle usage in the transportation sector has been on the rise. The engine is the essential component of the vehicle, and it needs proper lubrication to function efficiently. These lubricants must be replaced after a certain period of engine or motor running. As it contains polluted heavy particles, used petrol-engine oil (used-engine oil) is more hazardous to environment then crude oil. In addition to the heavy particle contamination, the unused oil contains significant amounts of polycyclic aromatic hydrocarbons (PAHs). Used-engine oil, especially used-petrol engine oil, negatively impacts the male reproductive system [27, 28] Waste engine oil can be used in concrete as a chemical additive. According to research, wasteengine oil meets the ASTM C494 standard's criterion for the type A water reduction additive in concrete [29].

Different researchers utilized used engine oil (UEO) in various kinds of composites. In various studies, UEO and used cooking oil were used to contrast and enhance the performance of asphalt pavement [30]. The use of UEO in asphalt pavement did not significantly increase the risk of low temperature cracking, according to research findings [31, 32]. On the other hand, wide number of studies on concrete with the addition of UEO have been reported. For a cost-effective method of dumping and environmental cleanup, the waste from the oil and fat industries was employed as an additive in concrete. UEO helps in the manufacturing of expanded clay aggregates used in research to create lightweight concrete [33-36]. Used petrol-engine oil (UPEO) and banana-fibers (BF) were combined to discover how they affected the concrete's workability. When UPEO was added, the workability increased [36]. Used engine oil can also degrade over time when used as a component in concrete. The degradation of used engine oil in concrete can be influenced by a variety of factors, such as exposure to air, moisture, heat, and chemical reactions with other materials in the concrete [53-55]. In addition to the physical changes that occur with the degradation of used engine oil in concrete, there may also be chemical changes that can result in the release of gases or other harmful substances. For example, the combustion of hydrocarbons in used engine oil can release carbon monoxide, nitrogen oxides, and other pollutants into the air. [56, 57]

To effectively mitigate the potential and harmful effects of degradation, it is importance to exercise great care and caution when it comes to utilizing used engine oil as an ingredient in concrete. One must ensure that the mixing and curing process is carried out in a meticulous and thorough manner, taking into account all possible factors that could contribute to the degradation of the material. According to this research, up to 9.4% mass of cement can be reduced and replaced with the UEO. It can be observed clearly that UEO enhanced the slump value of concrete. The initial setting was not affected but a delay observed in final setting time in comparison to normal plain concrete. The 28 days' compressive strength-



FIGURE 2.1: Effects of Addition of Waste Engine Oil (UEO) on Concrete's Properties [36]

was improved but a minor decrease was observed in 28 days' flexural strengths. The Figure 2.1 shows the influence of UEO in concrete [56]. UEO fulfills the ASTM C494 and it can be used as water reducing admixture. As per the results, the cement in concrete can be replaced up to 9.4% by mass and comparable properties can be achieved.

2.2.2 Banana Fibers

Various researchers have used natural fibers in composites in different ways and examined the impact that the adding of fibers was expected to produce. For the purpose of plastering the surface of structural components and walls, researchers used coconut fiber [37]. In a research study, an improvement was observed in out-plane lateral loading of the column with natural fibrous plaster on the column [25]. The researchers illustrated in a research that the treated fiber within the cement mortar improved the characteristics and durability properties of the composite [23]. Coconut fibers and coconut-fiber ropes were used in studies to enhance the resistance against the dynamic loading and significant improvement was noticed [38]. Banana fibers (BFs) are a readily available agricultural waste product worldwide, originating from banana farming. They have a variety of eco-friendly qualities, such as low density, light weight, low cost, water and fire resistance, and great tensile strength. Amorphous matrices of hemicelluloses and lignin make up the lignocellulosic materials that make up BFs, which are made of cellulose microfibrils embedded in a matrix. Both the amount of cellulose and the microfibril angle affect the fibres' mechanical characteristics. Banana fibres with a high cellulose concentration and low microfibril angle are capable of obtaining the requisite mechanical characteristics. Additionally, ligning are connected to hemicelluloses and considerably increase the lignocellulosic material's natural resistance to destruction. The water retting method, which includes submerging banana stems acquired from a farm in clean, potable water for six weeks, is used to extract BFs. The stems are then taken out of the water, loosened, and cleaned in a tank of clean water. The fibres are then cut to the proper length after being hand combed and sun-dried to further separate them. In a research work the influence of natural



FIGURE 2.2: Schematic Diagram of Fibre Bridging Effect [57]

fibers on self-compacting concrete was investigated [6]. Increment in the length of wheat straw fiber resulted in an increase in pullout peak load and pullout energy [39]. Natural fiber act differently in cement mortar and lime mortars. Natural fibers acted in favor of durability and the strength of the cement mortar [40]. A good mix design, more amount of fibers, and a large value of the water-cement ratio (for natural fibers) lead toward the great toughness and good strength of the composite [41]. The more quantity of banana fibers along with jute fibers in the composite resulted in an increase in the splitting-tensile strength, flexural strength, and impact resistance of the composite [13]. In comparison with other fibers, banana fiber has more average tensile-strength than coconut fiber, bamboo fiber, palm fiber, and sisal fiber [13]. The addition of banana fibers has caused an improvement in tensile strength of composite [42, 43] due to bridging effect as shown in **Figure 2.2**.

Chemical treatments can increase the strength of banana fibers (BFs) by eliminating impurities and improving wettability. There are several chemical treatments available to purify and strengthen natural fibers, with alkali treatment producing the best results. Water tends to be absorbed by BFs, especially in the first few hours after immersion. The surface roughness of the fibres is increased by alkali treatment, which enhances mechanical bonding and reduces water absorption. However, alkali treatment modifies the fibres' botanical constituents. Elbehiry and Mostafa (2018) evaluated the effects of sodium hydroxide at a concentration of 6 % for two hours at room temperature. Following a thorough rinsing in a water tank to get rid of any inactive reactions and make sure they were alkali-free, the fibres were dried for 24 hours at 80 °C. By soaking the fibres in NaOH solutions with concentrations ranging from 5 to 15 percent for three hours, Nor Azwin bt Ahad et al. (2019) evaluated the impact of the BF treatment. Since contaminants were observed on the surface of untreated fibers, the study concluded that the application of NaOH to the fibres from banana and coconut husks eliminated the contaminants. Alkaline treatment also increased the quantity of exposed cellulose on the surface and roughened the surface of the fibers, which can aid in interlocking the fibres and matrix.

A research study was conducted using the banana fiber bars as reinforcement. According to this study, the ability of concrete to resist cracking and spalling was increased [3]. Based on the findings of a comprehensive research study, it can be asserted that the incorporation of banana fiber bars as a reinforcing agent in concrete has been demonstrated to significantly enhance the material's ability to withstand cracking and spalling. Therefore, this study underscores the potential benefits of utilizing such natural fibers as an effective and sustainable solution for reinforcing concrete structures. The use of banana fibers significantly enhanced the resistance against cracking in the concrete beams [34]. This shows that addition of natural fiber contributed to reduction of depth of the concrete section by improving the flexural strength of concrete. The addition of banana and water hyacinth fibre to concrete was made using bio fillers, according to experimental research. The mechanical features of concrete—the maximum bearing load capacity, splitting-tensile strength, flexural strength, and compressive strength—as well as its physical characteristics—true density, bulk density, and water absorption—were improved by the introduction of water hyacinth and banana fibres. Banana fibre reinforced concrete was subjected to mechanical testing, and cement was substituted with banana leaf ash. Flexural strength and tensile strength were shown to have increased based on test findings [10].

2.3 Recoverable Concrete's Flaws with Waste Usage

There is no good building or construction material that can replace concrete in construction projects. In the construction industry, concrete is considered as a foundational element. Despite concrete's superiority, there are still a number of defects that need to be reduced or managed. The effects of flaws as weak in tension, cracking vulnerability, low impact resistance, and spalling, etc. The use of agricultural waste as dispersed reinforcement in concrete could, in some ways, reduce the impact of these flaws. The resistance to impact loading increased as a result of using agricultural waste and glass fiber reinforced polymer bars together [44, 45]. Utilizing both glass and steel fibers improved the energy absorption's ability to resist cracking again [46, 47]. The addition of jute fibers to concrete improved its resistance to impact loading and seismic performance [36, 48]. Concrete performed better when fine aggregates were swapped with the proper amounts of waste marble powder, that also provides an approach to sustainable development [36]. The behavior of mortar-free blocks to dynamic loads was studied using coconut fiber ropes generated from agricultural waste. The addition of glass fiber reinforced plastic waste increased the flexural strength of the concrete. Plastic waste and palm oil fuel ash combined to increase the ductility and energy absorption capabilities of concrete. In last 100 years, concrete production had imposed adverse ecological effects on environment. So, it is exerting contrary effects than benefits. Billions of tons of concrete are produced every year which emit carbon dioxide during mixing process and require large quantity of raw materials. The cement industry generated the total carbon dioxide of the world [11]. The agricultural and industrial wastes produced in urban and rural areas need proper disposal. The recycling of wastes to useful products is the solution of the crises of dumping wastes. The byproducts like silica fume, fly ash, granulated blast slag

and palm oil fuel ask had been used as sustainable construction materials [12]-[37]. These materials can be utilized in concrete for further use in different structural elements. Jute fiber was used by [46] as sustainable material for pavement application for freeze-thaw conditions. It was observed that the jute fiber reinforced concrete reduced the freeze-thaw effect. By the incorporation of jute fibers, the pavement with reduced thickness showed same results as compared with the pavement laid by the conventional concrete. The wheat straw has good properties to be used in cement composites as per studies conducted on the wheat straw fiber reinforced concrete. Wheat straw had been used by different researchers for application in concrete [13], [50]. A biobased composite i.e., wheat straw concrete was developed for the purpose of thermal insulation. The outcomes indicated that the biobased composite developed by using wheat straw showed excellent hygric properties. These properties can ensure the building's hygrothermal comfort [51]. Wheat straw was used in cement-based mortar for attaining sustainable cement mortar. Fiber length are used with fixed oil content. Natural fibers like banana fibers are locally available and used engine oil are also available. There is no proper method used to prepare thes material. Which are locally available in market with out treatment. Under the impact compression loading the specimen did not separate apart but only showed discrete cracking. The overall mechanical properties were enhanced with no change in thermal properties.

2.4 Governing towards Improvement in Concrete's Properties

Macro cracks are formed by the amalgamation of micro cracks. The properties and performance of concrete are affected by cracking and independent size of crack. The propagation of crack affects the concrete strength [49]. The phenomenon of cracking in concrete depends upon the properties of concrete [34]. The properties those influence the performance of the structure includes compressive strength, splitting-tensile strength, and flexural strengths. Splitting-tensile strength resists and control the cracking in concrete [39]. The durability and design of structure are affected up to some limits by above mentioned properties of the concrete. By making certain changes in production of concrete can alter the properties of concrete. This can be done, up to certain limits, by changing and/or introducing the ingredients to design consideration and durability considerations.

2.4.1 Material Behavior with Wastes

The ability of concrete to withstand abrasion, chemical attacks, and weathering over an extended period of time is commonly referred to as its durability. It is apparent that the mechanical properties of concrete play a significant role in durability in order to produce a construction material that is durable and extends the life of a structure. As concrete is a brittle material with weak flexural and tensile strength relative to compressive strength, durability of concrete depends on these properties as discovered in a research study [18]. Within 24 hours of concrete placement, the very first cracks were discovered on the concrete surface. At that time, composite does not have the ability to resist these microscopic cracks caused by ageing. durability of concrete means that ability of concrete to resist and with stand longer against weathering actions, checmical attacks and abrasion during the service. To obtain a durable construction material and service life of structure, it is obvious that mechanical properties of the concrete play important role to durability. In a research study, it was explored that the durability depend upon tensile and flexural properties of concrete are important as concrete is a brittle material, weak in flexural and tensile strength compared to compressive strength. The very first cracks found on the surface of the crack within 24 hours after placing of concrete [47]. At that time, composite does not have sufficient strength so it can resist these early age micro cracks. To avoid and resist these types of cracking there is need to add some additional constituents in concrete. The durability of concrete has inverse relation to linear shrinkage. On the other hand, high water absorption property of concrete has direct relation to durability of concrete and under severe environmental conditions, can cause corrosion of steel reinforcement inside concrete having cracks. The durability of structure also depends upon the

width of produced cracks and structural properties of concrete.

2.4.2 Strength Properties

There is need to add some additional ingredients to concrete in order to prevent and resist these types of cracks. Linear shrinkage has an inverse relationship with concrete durability. On the other hand, a high water absorption property of concrete has a direct correlation to concrete durability and can lead to corrosion of the steel reinforcement inside of cracked concrete under severe environmental conditions. Additionally, The durability of structure also depends upon the structural properties of concrete and width of produced cracks [50]

2.4.3 Water Absorption Properties

Compressive strength, splitting-tensile strength, and flexural strength are the key factors that determine the strength and design of structural members. Compared to tensile strength, concrete is much stronger under compression. Due to this, concrete's splitting-tensile and flexural strengths are given more importance in design structural members of concrete. In the different applications and according to strength aspects, splitting-tensile strength and flexural strength are given more importance than the compressive strength like in rigid pavement and beams, respectively. This splitting-tensile strength resist cracks caused by the shrinkage stresses. To obtain sustainable and high-strength material, banana fiber can enhance the mechanical properties of concrete. Water are absorb more in concrete of banana fiber mixed then PC concrete as pore available in banana fiber reinforced concrete.Water are absorb in concrete of banana fiber then PC concrete as pore available in banana fiber reinforced concrete. As water gets absorbed into the concrete matrix of banana fiber reinforced concrete, it fills up the pores present in the banana fibers, allowing for the effective transfer of stress between the fibers and the surrounding matrix of Portland cement concrete. Water absorbed more in concrete of banana fiber mixed then PC concrete as pore available in banana fiber reinforced concrete specimens.
2.4.4 Dynamic Properties

The researchers have investigated that concrete can be produced by replacing some amount of UPEO with cement comparable with PC. UPEO can be used as admixtures to reduce the cement content within the concrete result in minimize the cost production [13]. by improving concrete's mechanical and durability properties higher, the deterioration of concrete can be avoided. Natural fibers are locally available and used engine oil is very dangerous to environment as it is not dumped. We used both natural fiber and locally available these engine oil with out treatment. Natural fibers are locally available and used engine oil is very dangerous to environment as it is not dumpted. We used both natural fiber and locally available these engine oil with out treatment. used engine oil plain concrete specimens dynamic properties are good as compare to PC. Dynamic modulus of elasticity shown better values then dynamic modulus of rigidity. This highlights the importance of carefully assessing the intended use of concrete and selecting the appropriate strength criteria accordingly to ensure optimal performance and longevity of the structure. In the different applications and according to strength aspects, splitting-tensile strength and flexural strength are given more importance than the compressive strength like in rigid pavement and beams, respectively. This splitting-tensile strength resist cracks caused by the shrinkage stresses. To obtain sustainable and high-strength material, banana fiber can enhance the mechanical properties of concrete. The researchers have investigated that concrete can be produced by replacing some amount of UPEO with cement comparable with PC. UPEO can be used as admixtures to reduce the cement content within the concrete result in minimize the cost production [6]. by increasing concrete's mechanical and durability characteristics improved, the deterioration of concrete can be avoided

2.5 Summary

The discussion above comes to the conclusion that adding natural fibers and the admixture to concrete can enhance its durability-related properties. To prevent cracking and the growth of micro - cracks into macro cracks, it is necessary to improve mechanical and dynamic properties. It is clear from this chapter that banana fiber enhances the concrete's splitting-tensile strength and crack resistance. On the other hand, used petrol-engine oil can be used as a chemical admixture to concrete. Used petrol-engine oil improves the concrete's compressive strength to some level. The literature comes to the conclusion that used petrol engine oil has potential and can be used in combination with natural fires that have the same chemical (acid) nature as used petrol engine oil.

Chapter 3

Experimental Scheme

3.1 Background

Natural fibers are increasingly being used in concrete due to their low cost, simple handling, strong mechanical properties, easy availability, and environmentally friendly nature. In this research, used petrol engine oil is used as a chemical admixture in the production of concrete and banana fiber is used as reinforcement. The utilization of banana fibers and used petrol engine oil in various studies is discussed in detail in the previous chapter. However, the combined effects of used petrol engine oil and banana fibers have not yet been studied. Therefore, Slump cone test, dynamic test, mechanical test, tests for mass loss, linear shrinkage, and water absorption are taken into consideration. Broken specimens' fractured surfaces are also investigated at. These fiber length are used with fixed used engine oil content. Natural fibers like banana fibers are locally available and be cut into required length and used engine oil are locally available. The banana fiber having required cut lengths were commercially obtained from the market. Superplasticizer (Sika(R) ViscoCrete(R)-20 HE) was used in the UPC specimen to enhance strength of concrete. Superplasticizer was used 1% in UPC specimen by mass of cement. All these material cannot be treated. This chapter examines in great detail about the raw ingredients, fibre treatment, PC, UP, and UBFRCs mixing processes, casting process, and testing methodologies.

3.2 Raw Constituents

Ordinary Portland cement, Margalla crush, and readily available local sand are used to create normal plain concrete (PC). For the production of both fibers reinforced concrete and plain concrete, the aggregate size limit is 20 mm (FRC). While creating UPC and UBFRCs, UPEO is used as an admixture.the maximum permissible aggregate size is 20 mm. As for used petrol engine oil plain concrete (UPC) and used petrol engine oil banana fiber-reinforced concrete (UBFRC), used petroleum engine oil (UPEO) is incorporated as an admixture. It should be noted that all of the materials utilized in this study were sourced from local markets and were not subjected to any form of treatment or modification prior to use.

It should be noted that this research used commercially available UPEO. Since UPEO is considered as waste material industry, there is no information on its characterization. Banana fiber is used for the preparation of FRC. 3% BF content (by cement mass) was used. Varying lengths (3+5 cm, 3+7 cm, 5+7 cm, 3+5+7 cm)of fiber is used in the preparation of the FRC. Equal percentage of cut length are considered (for 3+5cm, 3+7cm, 5+7cm, 1.5% fiber for each length are considered; and for 3+5+7 cm, 1% fiber for each length are considered). The banana fibers having required cut lengths were commercially obtained from the market. UPEO is used within plain concrete to prepare used petrol-engine oil added plain concrete (UPC) and FRC. Superplasticizer (Sika® ViscoCrete®-20 HE) was used in the UPC specimen at 1% by mass of cement. The PC, UPC, and UBFRCs are prepared using normal-temperature tap water. Different type of specimens is manufactured using two different water-cement ratios. For the creation of PC and UPC, a water-cement ratio of 0.5 is used, and a ratio of 0.6 is used for all UBFRCs. increase english "According to a literature review, the BF (natural fibers) have a higher water absorption property, which has resulted in an increase in the water ratio for UBFRCs. Other raw material used for this research like used engine oil and admixture are arrange from local market. It should be noted that all of the materials used in this research were sourced from local markets and were not subjected to any form of treatment or modification prior to use. Loclly available materials like natural fiber and used engine oil are used.

3.3 Mix Design, Concrete Preparation, Fresh Properties, Casting of Specimens and Density Determination of Hard Concrete

The ratio of cement, sand, and aggregate in the mix design is 1:2:4 for the preparation of PC. Fixed amount 9.4% of UPEO by mass of cement is used for the preparation of UPC. For the preparation of used petrol engine oil banana fiber reinforced concrete (UBFRC), fixed proportion 3.0% and varying length (3+5 cm, 3+7 cm, 5+7 cm, 3+5+7 cm) BF are added to the mixture together with the UPEO. All the material is placed in the drum mixture for preparing the PC mix. Then water is added, in mixture machine, 30-45 seconds after start of rotating the mixture machine. The mixture machine is rotated for the five minutes. The slump cone test is performed after preparation of PC. For the preparation of the UPC, same procedure repeated with a change that used petrol engine oil is added one minute later after addition of the water. The rotating time for mixture is kept same five minutes as for the PC mix.

For the manufacturing of BFRC with varying length of (3+5 cm) and fixed content of 3.0% banana fiber by the mass of cement, the materials are placed in the form of layers to achieve the good mixing of fiber within the concrete. Three set of layers is used to' make a good mix of the UBFRC. One third set of layers of aggregates, sand, banana fibers and cements placed in the mixer machine. Then the second and third set of layers of aggregate, sand, banana fibers and cement are placed with the same approach. Then the mixture machine turned on to start rotating. And the two third water is added with the start of machine. Three minutes after of continuous rotation of mixer machine, remaining one third quantities of water and UPEO are added and mixture machine is kept rotating for further two minutes and slump cone test is performed to check the workability of the fresh UBFRC. Same approach was for the remaining the types of UBFRCs with varying length (3+7 cm, 5+7 cm, 3+5+7 cm) and fixed content (3.0%) of banana fibers. For the manufacturing of BFRC with varying length of (3+5 cm) and fixed content of 3.0% banana fiber by the mass of cement, the materials are placed in the form of layers to achieve the good mixing of fiber within the concrete. Three set of layers is used to' make a good mix of the UBFRC. One third set of layers of aggregates, sand, banana fibers and cements placed in the mixer machine. Then the second and third set of layers of aggregate, sand, banana fibers and cement are placed with the same approach. Then the mixture machine turned on to start rotating. And the two third water is added with the start of machine. Three minutes after of continuous rotation of mixer machine, remaining one third quantities of water and UPEO are added and mixture machine is kept rotating for further two minutes and slump cone test is performed to check the workability of the fresh UBFRC. Same approach was for the remaining the types of UBFRCs with varying length (3+7 cm, 5+7 cm, 3+5+7 cm) and fixed content (3.0%) of banana fibers

For the preparation of used petrol engine oil banana fiber reinforced concrete (UBFRC), fixed proportion 3.0% and varying length (3+5 cm, 3+7 cm, 5+7 cm, 3+5+7 cm) BF are added to the mixture together with the UPEO. To assess the workability or consistency of the manufactured precast concrete (PC), a slump cone test is conducted prior to pouring the concrete into molds. The ASTM standard C143/C143M-15a is followed, which specifies that the test be carried out with a slump cone having a bottom diameter of 200 mm (8 in), top diameter of 100 mm (4 in), and height of 300 mm (12 in), and a non-absorbent mold. The cone is filled with three equal volumetric layers of concrete, each of which is compacted 25 times with a tamping rod having a hemispherical end and a diameter of 16 mm (5/8 in) and length not exceeding 600 mm (25 in). The excess concrete is removed by striking off the tamping rod, and the surface is smoothed with screeding and rolling. The slump cone is lifted vertically upward, and the length of the reach over the concrete is measured using a ruler to determine the value of slump.

No standard test is available for evaluating the workability of fresh ultra-highperformance concrete (UPC) and ultra-high-performance fiber-reinforced concrete (UBFRC). Therefore, the same procedure and test standard used for PC is adopted for determining the workability of UPC and UBFRCs, to the best of the authors' knowledge. The relation between the observed slump values and determined hard densities are given in figure 3.1. Once the mixing is complete, the specimens are

cast, and mechanical tests are carried out at 28 days after curing. Prior to dynamic testing, tests such as water absorption, mass loss, and linear shrinkage are conducted. Once all the tests are completed, the results are compiled, and figures and tables are created to represent the findings. Subsequently, a conclusion is drawn based on the results. For measuring the densities, an average of two specimens is taken for each mix design. The volume of the beamlets is determined in terms of m3 by taking the internal volume of the moulds that are used for the casting of the beamlets. After the final setting time of the concrete, moulds are then removed and the mass in kilograms of each specimen is noted by using the weighing balance. The least count of weighing balance used for the determination of the masses is 5 grams. The densities are found by taking ratio of weight (kg) and volume (m^3) . Before casting of specimes will check fresh properties of concrete. Water to cement ratio and other values like slump value are also checked before casting of specimens same is case check density determination before casting. After mixing, casting of specimens other tests like mechanical tests are performed after 28 days and after curing. Water Absorption, Mass loss and Linear Shrinkage are performed before dynamic testing. After all these tests result are to be compile and figures and table are made of these result. After this all work conclusion are made and for further work recommendations are made. The fiber length are used with fixed oil content. Natural fibers like banana fibers are locally available and be cut into required length and used engine oil are locally available.

The workability of fresh concrete and density of hard used petrol engine oil plain concrete (UPC) and used petrol engine oil banana fiber-reinforced concrete (UBFRC) cannot be evaluated using standard tests, as far as the authors are aware. Therefore, the same procedure and test standard are used to determine the workability and densities of UPC and UBFRC.Following the mixing process, the specimens are cast, and various mechanical tests are conducted after 28 days of curing. Water absorption, mass loss, and linear shrinkage tests are then performed before the dynamic testing phase. Subsequently, all of the collected results are compiled and presented in figures and tables. Based on these findings, conclusions are drawn, and recommendations are provided for further work in this field.



FIGURE 3.1: a) Measuring the Value of Slump of UBFRC, b) Combined effect of UPEO and BF on the relation between slump of fresh concrete and density of hard concrete

		Addition o	fporcont	- aro		W/C	Slump of	Density of
Labeling	C:S:A	by cement	mass	-			fresh concrete	hard concrete
		UPEO	SP	BF	BF Length (cm)	(-)	(mm)	(kg/m^3)
PC	1:2:4	0	0	0	0	0.5	36	2356
UPC	1:2:4	9.4	1	0	0	0.5	40	2338
UBFRC3,5	1:2:4	9.4	0	3*	3+5	0.6	23	2264
UBFRC3,7	1:2:4	9.4	0	3*	3+7	0.6	19	2242
UBFRC5,7	1:2:4	9.4	0	3*	5+7	0.6	17	2185
UBFRC3,5,7	1:2:4	9.4	0	3**	3+5+7	0.6	13	2172

TABLE 3.1: Mix Design, Specimen Labeling, Slump of Fresh Concrete, and Density of Hard Concrete

* 1.5% for each length text ** 1.0% for each length

3.4 Testing Methodology

All tests conducted in this study are summarized in Table 3.2 and explained in following sections. The mechanical tests (i.e. destructive testing) are conducted after the dynamic and miscellaneous tests. In miscellaneous tests, selected specimens of all combinations were exposed to different exposure conditions as per their standards in order to check the maximum degradation in specimens strengths. **Figure 3.2** shows different test setups.



FIGURE 3.2: Tests Setup; a) Dynamic Test, and b) Mechanical Properties

3.4.1 Dynamic Tests

Dynamic test is performed before the destructive (mechanical) testing of the specimens as per ASTM 215-14. Response frequencies lateral (RFL), response frequencies transverse (RFT) and response frequencies rotational (RFR) are determined with the help of hammer and accelerometer. The test is conducted on both cylinder and beamlets. For determining the RFL, the accelerometer is attached to once side of cross section of cylinders and beamlets while a strike of hammer is given to the opposite side of the cross section of specimens. The accelerometer observes the frequencies and transfer the record of these frequencies to the computer attached with it. The procedure of RFT and RFR attaching the accelerometer and strike location of hammer is different for cylinders and the beamlets. In case of cylinders, for RFT, the accelerometer is attached at side showing face of length of cylinder at least 25 cm away before the edge. Then a strike of hammer is given at same side showing face on the center of cylinder's length.

For RFR, the accelerometer is attached at top showing face of length of cylinder with same space from the edge like RFT. The strike is given at perpendicular accelerometer on opposite edge of the cylinder length. In case of the beamlets, for RFT determination, accelerometer attached at one side of length at same margin, used for cylinders, on length of beamlets from edge. Strike of hammer is given at center of length of same side at which accelerometer is attached. For RFR, the accelerometer is attached at top corner of rectangle (side face of the beamlet). A strike is given at other side bottom corner of same side of rectangle in such a way that line joining the point of hammer's strike and accelerometer make the diagonal of the rectangle. From these observed frequencies, damping ratio, dynamic modulus of elasticity, the dynamic modulus of modulus of rigidity and poison's ratios are calculated. These calculated properties support to understand the behavior and resistance of PC, UPC, and all types of UBFRC against the dynamic loading. These properties are key to the design of structure undergoing the dynamic loadings and earthquake

3.4.2 Miscellaneous Tests (Water Absorption, Linear Shrinkage, and Mass Loss)

To calculate the water absorption properties of PC, UPC, and UBFRCs, the ASTM C642 standard is followed. First of all, specimens are dried in the oven and then these dried specimens are placed, at room temperature, in water. This method is used to determine the water absorption property of all types of specimens. For the evaluation of linear shrinkage, ASTM C157 / C157M-08 is followed by observing and measuring the variations in the length of specimens (OPSS standard LS-435). For this purpose, a line of 6 inches is marked as a reference on the length of specimens before conducting the test. The variation of the length is

Test	Standards / References	Parameters Considered for the Study
1. Dynamic Properties	ASTM 215-14	Resonant frequency longi- tudinal (RFL), Resonance frequency transverse (RFT), Resonance frequency torsional (RFR), damping ratio (), Poisson ratio, dynamic modu- lus of elasticity and dynamic modulus of rigidity (Ed)(Rd).
2. Miscellaneous		
a. Water Absorption	ASTM C642- 13	Water absorption (%)
b. Linear Shrinkage	ASTM C157	Linear shrinkage (percentage decrease),
c. Mass Loss	ASTM C157M-08	Mass loss by gradually increas- ing temperature
3. Mechanical Properties		
a) Compressive Properties	ASTM C39	Stress-strain curves, compres- sive strength (CS), modulus of elasticity (MOE) compres- sive pre-crack energy absorp- tion (CE1).

TABLE 3.2: Standards Testing and Studied Parameters	
---	--

Test	Standards / References	Parameters Considered for the Study
		Compressive toughness indexes, compressive post-crack energy absorption (CE2), and com- pressive total energy absorption (CTE) (CTI).
b) Splitting Tensile Properties	· ASTM C496M-02	Load-deformation curves, splitting-tensile toughness in- dexes, splitting-tensile strength (SS), splitting-tensile pre- crack energy absorption (SE1), splitting-tensile post-crack energy absorption (SE2), and splitting-tensile total energy absorption (STE) (STI).
c) Flexural Properties	ASTM C78	Load-deflection curves, flexu- ral toughness indexes, flexural strength (FS), flexural pre-crack energy absorption (FE1), flexu- ral post-crack energy absorption (FE2), and flexural total energy absorption (FTE) (FTI).
4. Role of Fibers in Con- crete	- Affan, M (2019)	Broken surfaces of specimen, failure mechanism of fibers, and bonding of fiber with the sur- rounding matrix

measured after following the procedure of standard. measured after following the procedure of standard.

The linear shrinkage is then measured by taking percentage difference of marked length before and after the test procedure. ASTM C157M-08 is used for the determination of mass loss in PC, UPC and UBFRCs.

After following the test procedure, various result are obtained related to shrinkage water ratio and mass loss. As pore are increases due to fibers. When fibers increasing in contents are with length. Shrinkage are less as compare to other miscelenious values like mass loss. Variations and shrinkage in the reference line are marked before being evaluated. Each type of concrete mix specimen is placed in a high-temperature heating oven. After following the test procedure, various result are obtained related to shrinkage water ratio and mass loss. As pore are increases due to fibers. When fibers increasing in contents are with length. Shrinkage are less as compare to other miscelenious values like mass loss. Variations and shrinkage in the reference line are marked before being evaluated. Each type of concrete mix specimen is placed in a high-temperature heating oven. The temperature is raised from 20°C to 100°C at the rate of increase of 3°C per minute and maintained at 100°C for one hour. This is done to obtain more realistic data. Then specimens are cooled down with the same rate of decrease in temperature at 3°C to avoid thermal cracking. The temperature is raised from 20°C to 100°C at the rate of increase of 3°C per minute and maintained at 100°C for one hour. This is done to obtain more realistic data. Then specimens are cooled down with the same rate of decrease in temperature at 3°C to avoid thermal cracking.

3.4.3 Mechanical Tests

a) Compression test

A servo-hydraulic testing machine (STM) is used for the determination of the compressive strengths of PC, UPC, and UBFRCs. The test is performed according to ASTM C39 on cylinders of PC, UPC, and UBFRCs.

In this test properties are determined to include compressive strength (CS), compressive behavior, compressive pre-crack (CE1) and post-crack energy (CE2), compressive total absorbed energy (CTE), and the compressive toughness index (CTI) of PC, UPC, and UBFRC. To distribute the load uniformly throughout the cylinder the capping of the cylinder is done with the plaster of paris.

b) Splitting Tensile Test

ASTM C496M-02 standard is used for the splitting-tensile test.

The same machine STM is used for performing the test. The test is performed on the cylinders of PC, UPC, and UBFRCs. The capping of cylinders is not required in the case of the splitting-tensile test. From this test, load-deformation curves, splitting-tensile strength (SS), splitting-tensile pre-crack energy absorption (SE1), splitting-tensile post-crack energy absorption (SE2), splitting-tensile total energy

absorption (STE), and splitting tensile toughness indexes (STI) are calculated.

c) Flexural Test

The flexural test is performed on the basis of standards ASTM C78. Flexural test in performed in Servo-Hydraulic Testing Machine. The three-point loading mechanism is adopted. The test is performed on the beamlets of PC, UPC, UBFRCs. Two specimens are prepared for each length, and the average result is taken after testing. The studied parameters in this test are load-deflection curves, flexural strength (FS), flexural pre-crack energy absorption (FE1), flexural post crack energy absorption (FE2), flexural total energy absorption (FTE), and flexural toughness indexes (FTI).

3.4.4 Visual Inspection of Crack Surface

After performing the mechanical testing, the fractured surfaces of broken specimens are examined carefully. In this examination, fiber breakage, pullouts, and bridging effect due to fibers are investigated. For this purpose, microanalysis is done carefully over the broken surfaces of the tested samples. From the fracture surfaces, mixing of all the ingredients can be observed, either it is a good mix or not. It is through achieving a proper mixing of ingredients that the desired properties can be obtained. Good mixing of ingredients results in achieving the desired property, otherwise, it can lead to digression of the property instead of increasing it. The basic purpose of this investigation is to elaborate the failure mechanism of fibers and the bonding of the fibers with the surrounding matrix. Following mechanical testing, the shattered surfaces of the damaged specimens are closely inspected. This investigation looks at fibre breakage, pullouts, and abridging effects caused by fibres. Microanalysis is carefully performed across the damaged surfaces of the analysed materials for this reason. The blending of all the materials, whether it is a good combination or not, can be seen on the fracture surfaces. It is through achieving a proper mixing of ingredients that the desired properties can be obtained. Good mixing of ingredients results in achieving the desired property, otherwise, it can lead to digression of the property instead of in

-creasing it. It is a good mix or not.

If ingredients are mixed properly, the intended characteristic is produced; if not, the property may actually decrease instead of increase. Fibers after cracking of concrete material have the ability to control brittle failure of concrete. Thus by using fiber brittle nature of fiber to ductile nature. And these type of fiber are locally available. but strength of concrete is reduced specially compressive strength. The main goal of this inquiry is to clarify the connection between the fibres and the surrounding matrix and the failure mechanism of the fibres.

3.5 Summary

PC, UPC, and UBFRCs are prepare using the 1:2:4 mix design, which is the most widely used. The UBFRCs used 0.6 w/c whereas the PC and UPC used 0.5 w/c. For the preparation of UPC and UBFRCs, a fixed amount of 9.4% of UPEO by mass of cement is used. For the preparation of used petrol engine oil banana fiber reinforced concrete (UBFRC), fixed proportion 3.0% and varying length (3+5 cm, 3+7 cm, 5+7 cm, 3+5+7 cm) BF are added to the mixture together with the UPEO. Total number of 36 specimens are prepared in which 24 are cylinders and 12 are the beamlets. ASTM standards are followed in slump, dynamic, mechanical, and miscellaneous tests of PC, UPC, and UBFRCs. The evaluated results of each corresponding tests are discussed in detail in next chapter (i.e., chapter 4).

Chapter 4

Results and Analysis

4.1 Background

PC, UPC, and UBFRCs are prepare using the 1:2:4 mix design, which is the most widely used. The UBFRCs used 0.6 w/c whereas the PC and UPC used 0.5 w/c. For the preparation of UPC and UBFRCs, a fixed amount of 9.4% of UPEO by mass of cement is used. For the preparation of used petrol engine oil banana fiber reinforced concrete (UBFRC), fixed proportion 2.0% and varying length (3+5 cm, 3+7 cm, 5+7 cm, 3+5+7 cm) BF are added to the mixture together with the UPEO. This chapter is based on comprehensive findings that were attained following testing of each PC, UPC, and UBFRC specimen.

4.2 Dynamic Properties

To evaluate the combined effect of UPEO and BF on the properties of concrete specimens, dynamic properties are investigated. ASTM C215-14 is used to determine these dynamic properties of concrete (PC) specimens. The same methods are used to determine the dynamic properties of UPC and UBFRCs as there is no specific standard available for this purpose. Due to the absence of a specific standard similar methods are often utilized. Table: 4.1 shows the investigated

values is taken to achieve dynamic properties of the PC, UPC, and UBFRCs. For this, an average of two appropriate results of corresponding dynamic properties. The damping ratio (ζ) of UPC is reduced by 6.42% and 21.88% in cases of cylinders and beamlets, respectively, as compared with PC. As comparing the values with PC, the damping ratios of UBFRC3+5, UBFRC3+7, UBFRC5+7, and UBFRC 3+5+7 are increased by 18.02%, 20.27%, 25.4%, 29.48% in the case of cylinders respectively. In the case of beamlets, the damping ratio of UBFRC3+5, UBFRC3+7, UBFRC5+7, and UBFRC 3+5+7 are increased by 1.52%, 6.25%, 9.3% and 17.02% in comparison with the damping ratio of PC respectively. Increment/decrement in demping ratio is directly releted to resistence against dynamic loading. The resistance against the dynamic loading has been reduced by the additive influence of the UPEO. On the other hand, the addition of the BF has improved the resistance against the dynamic loading impact in comparison to the simple plain concrete. In the case of cylinders, the impact of dynamic modulus of elasticity (Ed) is enhanced by 6.15% the 9.4% addition of UPEO in concrete. Contrarily the reduction is observed in values of Ed of UBFRC3+5, UBFRC3+7, UBFRC5+7, and UBFRC 3+5+7 by 7.02%. 10.24%, 13.66% and 15.82% respectively.

The dynamic modulus of rigidity has shown better values in the case of the cylinder than the beamlets. From this experiment, it is observed that the addition of the BF in concrete has improved the dynamic properties of the fiber reinforced concrete than that of plain concrete. The increments are observed in both types, cylindrical test specimens and beamlets test specimens. This is an indication that the members made of UPEO and BF can resist and withstand more against the lateral loading in either it is a cylindrical or a beamlet. These improved properties are the indications that the occupancy of some portion of BF in UBFRC can sustain more against impact loading and may be enhance the durability against earthquake loading as compared to that of plain concrete. In the case of the cylinder instead of the beamlets, the dynamic modulus of rigidity has shown better values. According to the results of this experiment, addition of BF to concrete improved the dynamic properties of fiber reinforced concrete more than plain concrete. Both types of test specimens beamlets and cylindrical show the increments. This is an indication that UPEO and BF members can withstand and resist lateral loads more effectively, whether they are cylindrical or beamlet. These improved properties are the indications that the occupancy of some portion of BF in UBFRC can sustain more against impact loading and may be enhance the durability against earthquake loading as compared to that of plain concrete. This is an indication that UPEO and BF members can withstand and resist lateral loads more effectively, whether they are cylindrical or beamlet. These improved properties are the indications that the occupancy of some portion of BF in UBFRC can sustain more against impact loading and may be enhance the durability against earthquake loading as compared to that of plain concrete.

The dynamic modulus of rigidity has shown better values in the case of the cylinder than the beamlets. From this experiment, it is observed that the addition of the BF in concrete has improved the dynamic properties of the fiber reinforced concrete than that of plain concrete. The increments are observed in both types, cylindrical test specimens and beamlets test specimens. This is an indication that the members made of UPEO and BF can resist and withstand more against the lateral loading in either it is a cylindrical or a beamlet. These improved properties are the indications that the occupancy of some portion of BF in UBFRC can sustain more against impact loading and may be enhance the durability against earthquake loading as compared to that of plain concrete. The dynamic modulus of rigidity has shown better values in the case of the cylinder than the beamlets. From this experiment, it is observed that the addition of the BF in concrete has improved the dynamic properties of the fiber reinforced concrete than that of plain concrete. The increments are observed in both types, cylindrical test specimens and beamlets test specimens. This is an indication that the members made of UPEO and BF can resist and withstand more against the lateral loading in either it is a cylindrical or a beamlet. These improved properties are the indications that the occupancy of some portion of BF in UBFRC can sustain more against impact loading and may be enhance the durability against earthquake loading as compared to that of plain concrete. This is an indication that the members made of UPEO and BF can resist and withstand more against the lateral loading in either it is a cylindrical or a beamlet.

Concrete	Parameters						
Specimen Type	RFL	RFT	RFR	ζ	Ed	Rd	Poisson Ratio
	(Hz)	(Hz)	(Hz)	(%)	(GPa)	(GPa)	(-)
Cylinders*							
PC	3417 ± 44	3417 ± 45	3298 ± 22	$3.88 {\pm} 0.104$	$4.30 {\pm} 0.040$	$4.40 {\pm} 0.165$	$0.52{\pm}0.005$
UPC	3452 ± 23	3387 ± 48	3392 ± 23	$1.68 {\pm} 0.058$	$4.30 {\pm} 0.019$	$4.01 {\pm} 0.045$	$0.47 {\pm} 0.008$
UBFRC3+5	3506 ± 23	$3232{\pm}23$	$3245 {\pm} 44$	$3.83 {\pm} 0.056$	$3.70 {\pm} 0.015$	$4.91 {\pm} 0.006$	$0.53 {\pm} 0.005$
UBFRC3+7	3535 ± 89	$3255{\pm}42$	$3263{\pm}65$	$3.73 {\pm} 0.203$	$3.70 {\pm} 0.014$	$4.80 {\pm} 0.022$	$0.51 {\pm} 0.015$
UBFRC5+7	3478 ± 22	3384 ± 82	3392 ± 28	$3.71 {\pm} 0.211$	$4.10 {\pm} 0.035$	$4.00 {\pm} 0.033$	$0.49 {\pm} 0.003$
UBFRC3+5+7	3412 ± 52	3333 ± 66	$3205{\pm}62$	$3.29 {\pm} 0.017$	$4.00 {\pm} 0.023$	$3.70 {\pm} 0.748$	$0.46{\pm}0.012$
Beamlets**							
PC	3395 ± 88	$3231{\pm}67$	$3385 {\pm} 26$	$1.76 {\pm} 0.022$	21.3 ± 1.785	$26.6 {\pm} 0.645$	$0.60{\pm}0.004$
UPC	$3488{\pm}65$	3334 ± 66	3292 ± 35	$1.77 {\pm} 0.051$	22.3 ± 1.151	26.3 ± 0.412	$0.58 {\pm} 0.003$
UBFRC3+5	3245 ± 25	3340 ± 38	3332 ± 57	$3.35 {\pm} 0.052$	$21.6 {\pm} 0.486$	27.5 ± 0.002	$0.61 {\pm} 0.010$
UBFRC3+7	$3288 {\pm} 38$	$3418{\pm}28$	$3298{\pm}86$	$2.83{\pm}0.057$	$22.3 {\pm} 0.745$	$25.9 {\pm} 0.112$	$0.57 {\pm} 0.018$
UBFRC5+7	3312 ± 78	3408 ± 98	3357 ± 24	$2.78 {\pm} 0.033$	$22.5 {\pm} 0.552$	$24.6 {\pm} 0.052$	$0.54{\pm}0.014$
UBFRC3+5+7	$3535{\pm}89$	$3360{\pm}55$	3205 ± 36	$2.72 {\pm} 0.014$	$21.9 {\pm} 0.912$	24.3 ± 0.122	$0.55 {\pm} 0.012$

TABLE 4.1: Dynamic Properties of PC, UPC, and UBFRCs

4.3 Miscellaneous Properties

Water absorption, mass loss, and linear shrinkage are all important properties used to determine the suitability of materials for various applications. Water absorption is a measure of a material's ability to absorb water, and it can affect the durability and strength of the material. Materials with high water absorption may be more susceptible to damage from environmental factors such as freezing and thawing cycles or exposure to moisture. Mass loss is another property that is often used to evaluate materials. This property measures the amount of weight loss that occurs when a material is exposed to a specific temperature or environment. It can be an indicator of the material's thermal stability or its ability to resist degradation from exposure to chemicals or other environmental factors. Mass loss is an important property. Linear shrinkage is a measure of how much a material will shrink or expand when it is exposed to heat or moisture. This property can be important in applications where tight tolerances are necessary, such as in the manufacturing of electronic components or in the construction of buildings. Miscellaneous properties are shown in **Table 4.2**.

4.3.1 Water Absorption,

Water absorption is an important property of concrete as it affects its durability and strength. Water absorption process occurs through capillary action, and usually calculated as the absorbed water total mass divided by the actual mass of specimen's after oven drying. In this study, the water absorption percentage of PC, UPC, and UBFRCs were measured, and the results are presented in Table 4.3 shows the values of water absorption percentage of plain concrete (PC), used petrol engine oil plain concrete (UPC), and UBFRCs; these are 2.53, 2.76, 3.05, 3.95, 3.98, and 3.84 respectively. The water absorption for the UBFRC3+7 and UBFRC5+7 are relatively higher. And the water absorption values are increased with the increase in the length of fiber. So, this shows that the water absorption property of concrete increased with the increase in the length of BF.

4.3.2 Mass Loss

The high temperature oven is used for performing the mass loss test. The method described by ASTM standard C157M-08 is adopted to measure linear shrinkage and mass loss in PC, UPC, and UBFRCs. The values of linear shrinkage PC, UPC and UBFRCs are 0.188, 0.166, 0.122, 0.098, 0.092, and 0.095 respectively.

TABLE 4.2: Water Absorption, Linear Shrinkage, and Mass Loss of PC, UPC, and UBFRCs

Concrete	Water Absorption	Linear Shrinkage		Mass Lo	SS
Туре	(%)	(%)	0	0	0
			$50^{\circ}\mathrm{C}$	$75^{\circ}\mathrm{C}$	100°C
PC	$1.55 {\pm} 0.5$	0.188	-0.034	-0.037	-0.104
UPC	$1.10{\pm}0.2$	0.166	-0.044	-0.048	-0.125
UBFRC3+5	$3.10{\pm}0.5$	0.122	-0.053	-0.067	-0.147
UBFRC3+7	$2.7 {\pm} 0.5$	0.098	-0.059	-0.071	-0.156
UBFRC5+7	$2.65{\pm}0.6$	0.092	-0.069	-0.078	-0.185
UBFRC3+5+	7 3.95 ± 0.2	0.095	-0.070	-0.081	-0.195

4.3.3 Linear Shrinkage

The measurement of linear shrinkage in PC, UPC, and UBFRCs was conducted using the ASTM standard C157M-08. The values of linear shrinkage PC, UPC and UBFRCs are 0.188, 0.166, 0.122, 0.098, 0.092, and 0.095 respectively. The data revealed that an increase in fiber length led to a decrease in the value of linear shrinkage, indicating that BF can resist linear shrinkage. Specifically, the results showed that increasing the length of the fibers led to a decrease in the value of linear shrinkage, While UBFRCs still exhibited some linear shrinkage, it was less with a higher content of BF, suggesting that the addition of BF could potentially reduce the risk of cracks in concrete. The addition of UPEO in UPC resulted in a decrease in the value of linear shrinkage in comparison to PC, implying that the presence of UPEO reduced voids in concrete.

4.4 Mechanical Properties

4.4.1 Compressive Properties

Figure 4.1 displays stress-strain graphs for plain concrete (PC), UPEO-modified plain concrete (UPC), and various ultra-high-performance fiber reinforced concretes (UBFRCs) containing different amounts of basalt fiber (BF). UPC exhibits the highest compressive strength (CS) among all specimens, with an increase of 2.17% compared to PC. However, the CS of UBFRCs with higher amounts of BF decreases, with reductions of 6.23%, 23%, 40% and 31% for UBFRC3+5, UBFRC3+7, UBFRC5+7 and UBFRC3+5+7, respectively. This suggests that UPEO enhances the CS of concrete while BF reduces it. **Table 4.3** shows the modulus of elasticity (MOE) and various absorbed energies and toughness indices under compression for all specimens. The CS of concrete has increased when UPEO is added in normal concrete. On the other way around, the addition of BF has caused in negative effect. Hence, CS is decreased with an increase in length of BF. The MOE of UPC is 1.08% higher than that of PC, but the MOE of UBFRCs decreases as the length of BF increases.

Table 4.3 shows the values of compressive pre-crack absorbed energy (CE1), compressive post crack absorbed energy (CE2), compressive total absorbed energy (CTE), and compression toughness index (CTI). All these types of compressive absorbed energies are calculated as per criteria and method described by in the research studies. There is reduction of 33% 40% and 60% in the values of CE1 for UPC, UBFRC3+7 and UBFRC5+7 respectively, when compared with PC. On the other hand, the respective CE2 is increased by 20%, 11.11%, 33.88%, 36.5%, and 40.3% as compared to PC. There is reduction of 33%, 40% and 60%, in the values of CE1 for UPC, respectively, when compared with PC CE2, and CTI generally increase as the length of BF increases, with the largest CE2 observed in UBFRC3+5+7. These improvements in absorbed energies and toughness indices are due to the addition of BF, and further improvements are achieved through the incorporation of UPEO. Figure 4.1b shows the relationship between stress and strain for all specimens under compressive loading.



25 PC 20 UPC 3+5 **Stress (MPa)** 10 3+7 5+7 3+5+7 5 0 0.025 0.005 0.01 0.015 0.02 0.03 0.035 0.04 0 Strain (-) b)

FIGURE 4.1: Compression Behavior ; a) Typical Tested Compression Specimens and b) Stress Strain Curves

4.4.2 Splitting Tensile Properties

Based on the information provided, it seems that the addition of banana fibers (BF) has a positive effect on the splitting-tensile strength of concrete. The loaddeformation curves presented in **Figure 4.2** show that specimens of plain concrete (PC) and concrete with used petrol engine oil(UPC) failed suddenly after reaching their maximum load, while the addition of 3% banana fibers to UPC with varying length improved its splitting-tensile strength and post-crack energy absorption. Moreover, increasing the length of BF led to an improvement in the splitting-tensile strength and energy absorption of the UBFRC (used banana fiber-reinforced concrete) specimens. Also, it has absorbed some energy after appearance of first crack and did not failed suddenly as compared to PC and UPC. The addition of high length BF has absorbed more post-crack energy along with more splitting-tensile strength. **Figure 4.2** shows comparison between typical failure of PC and UBFRC under maximum splitting-tensile loading.

By admixing (3+5 cm) of banana fibers along with same content of UPEO used in UPC, the splitting-tensile strength is improved up to some extent and, also, resisted the some loading after appearance of first crack. This has proved that the addition of banana fibers has improved the splitting tensile strength. Also, it has absorbed some energy after appearance of first crack and did not failed suddenly as compared to PC and UPC. The addition of high length BF has absorbed more post-crack energy along with more splitting-tensile strength. Figure 4.2a shows comparison between typical failure of PC and UBFRC under maximum splitting-tensile loading. Table 4.3 summarizes the splitting-tensile properties of the different concrete specimens, including splitting-tensile strength (SS), precrack absorbed energy (SE1), post-crack absorbed energy (SE2), total absorbed energy (STE), and toughness index (STI). The presence of BF is reported to enhance the splitting-tensile strength of concrete. It can be noted that UBFRC5+7 has the maximum load as shown in **Figure 4.2b**. Also, after the maximum load, the UBFRCs have shown load-carrying capability due to the bridging effect of BF. Under the section of splitting-tensile properties, splitting-tensile strength (SS), splitting-tensile pre-crack absorbed energy (SE1),





a) 60 PC UPC 50 3+5 3+7 Load (kN) 5+7 40 3+5+7 30 20 10 0 0.2 0.4 0.6 0.8 1 1.2 0 **Deformation (mm)** b)

FIGURE 4.2: Tension Behavior ; a) Typical Tested Tension Specimens and b) Load Deformation Curves

4.4.3 Flexural Properties

Figure 4.3 shows the relationship between load-deflection curves of PC, UPC, UBFRC3+5, UBFRC3+7, UBFRC5+7, and UBFRC 3+5+7 samples under flexural loading. UBFRC5+7 has resisted the maximum flexural load as shown in Figure 4.3b. It may be observed that maximum flexural strength in the case of the UBFRC5+7 as shown in **Table 4.3**. The more deflection is experienced in UBFRCs than the PC and UPC. In the section on flexural properties, Table 4.2 presents data on flexural strength (FS), flexural pre-crack absorbed energy (FE1), flexural post-crack absorbed energy (FE2), flexural total absorbed energy (FTE), and flexural toughness index (FTI). This behavior is primarily due to the bridging effect caused by the fiber reinforcement. The FS is increased and has shown maximum value in the case of UBFRC5+7 as compared with the other FS values of other specimens. This increase in FS is caused by the addition of the optimum value of BF in concrete. Beyond this, the value of FS is reduced. Table 4.3 presents data on flexural pre-crack absorbed energy (FE1), flexural post-crack absorbed energy (FE2), flexural total energy absorption (FTE), and flexural toughness indexes (FTI) for different concrete types, including UBFRC with a high content of BF. Compared to PC, the flexural pre-crack absorbed energy of UBFRC2.0 decreased by 15.33% due to the presence of UPEO. The reduction in FE1 for UPC, compared to PC, is also attributed to the presence of UPEO. PC samples did not exhibit any flexural post-crack absorbed energy as they broke into two pieces at the peak load. However, the addition of UPEO led to an increase in the concrete slump value, while the addition of BF decreased the concrete slump value. The efficiency and performance of concrete are affected by these loadings, including FS properties. While All types of UBFRCs have shown some of the FE2 This means the UBFRC may sustain better than PC and UPC after the cracks appear in the concrete. BF has resisted the sudden failure of under Flexural loading concrete comparing to those of PC and UPC. While All types of UBFRCs have shown some of the FE2 This means the UBFRC may sustain better than PC and UPC after the cracks appear in the concrete. BF has resisted the sudden failure of under Flexural loading concrete comparing to those of PC and UPC.



a)



FIGURE 4.3: Flexural Behavior ; a) Typical Tested Flexural Specimens and b) Load Deflection Curves

TABLE 4.3: Compressive, Splitting-Tensile, and Flexural Properties of PC, UPC, and all UBFRCs

	Param	neters															
Concrete	Compression Properties						Splitting-Tensile Properties				Flexural Properties						
Туре	CS	MOE	CE1	CE2	CTE	CTI	\mathbf{SS}	SE1	SE2	STE	STI	\mathbf{FS}	Δ	FE1	FE2	FTE	FTI
	(MPa)	(GPa)	$\left(\mathrm{MJ}/\mathrm{m}^{3}\right)$	$\left(\mathrm{MJ/m^{3}}\right)$	$\left(\mathrm{MJ/m^{3}}\right)$	(-)	(MPa)	(Nm)	(Nm)	(Nm)	(-)	(MPa)	(mm)	(Nm)	(Nm)	(Nm)	(-)
BC	23.72	27.77	0.1	0.068	0.167	1.68	1.5	9.92	0	9.92	1	3.22	0.32	0.78	0	0.78	1
rC	± 0.5	± 1.4	± 0.02	± 0.01	± 0.01	± 0.02	± 0.1	± 0.2	0 ± (± 0.2	T	± 0.2	± 0.1	± 0.1	0	± 0.1	1
UPC	12.73	20.33	0.071	0.052	0.124	1.73	1	4.39	0	4.39	1	2.9	0.51	1.17	0	1.17	1
	± 0.2	± 0.7	± 0.03	± 0.01	± 0.01	± 0.03	± 0.2	± 0.1		± 0.1		± 0.1	$\pm 0.1 \pm 0.1 \pm 0.2 =$	± 0.2	2		
UBFRC3+5	11.65	19.45	0.062	0.173	0.235	3.79	0.96	6.35	9.07	15.4	2.43	2.64	0.5	1.14	0.32	1.46	1.28
	± 0.4	± 1.2	± 0.02	± 0.01	± 0.02	± 0.04	± 0.1	± 1.2	± 1.2	± 1.2	± 0.2	± 0.1	± 0.1	± 0.1	± 0.1	± 0.1	± 0.1
UBFRC3+7	12.09	19.81	0.055	0.142	0.197	3.57	1.5	9.66	14.3	24.1	2.48	3.71	0.57	2.18	0.44	2.63	1.2
	± 0.2	± 1.2	± 0.02	± 0.01	± 0.01	± 0.02	± 0.2	± 1.2	± 1.3	± 1.5	± 0.1	± 0.2	± 0.1	± 0.2	± 0.1	± 0.2	± 0.1
UBFRC5+7	6.8	14.9	0.041	0.1	0.141	3.43	1.48	10.4	17.1	27.4	2.65	4.21	0.87	3.01	0.3	3.31	1.1
	\pm .9	± 1.2	± 0.02	± 0.01	± 0.01	± 0.02	± 0.2	± 1.3	± 1.5	± 2.4	± 0.2	± 0.2	± 0.1	± 0.2	± 0.1	± 0.2	± 0.1
UBFRC3+5+7	8.52	16.7	0.029	0.155	0.184	6.25	1.66	13.7	16.8	30.5	2.23	3.63	0.56	1.98	0.36	2.4	1.18
	± 1.0	± 0.3	± 0.01	± 0.01	± 0.01	± 0.01	± 0.2	± 0.2	± 1.3	\pm 3.2	± 0.2	± 0.2	± 0.1	± 0.1	± 0.1	± 0.2	$\pm~0.1$

49

4.5 Fiber Breakage and Pullout

Figure 4.4 shows broken surfaces of fibrous concrete after mechanical loading. The failure pattern and surfaces of Ultra-High-Performance Fiber-Reinforced Concretes (UBFRCs) are found to be different from that of plain concrete (PC). Fractured surface images are used to study the bonding between banana fibers and the surrounding concrete. The images reveal clear fiber pullout, indicating a strong bonding between the fibers and the matrix. The broken surfaces of the tested specimens also exhibit very few and extremely small voids, which is an indication of proper mixing of concrete ingredients.

After mechanical testing, UBFRC specimens are broken into small pieces that are attached to other broken particles. The observed attachment is believed to be the outcome of the robust bridging impact of the banana fibers present within the composite material, whereby the fibers act as reinforcement agents, spanning across any gaps or fissures in the material and enhancing its overall mechanical stability and strength. which is not observed in normal plain concrete. The presence of fibers also causes fiber pullout at the location of the fracture surface instead of fiber breakage, indicating that the UBFRC specimen can withstand loading longer than PC. After the failure of the UBFRC specimen, it continues to resist loading due to the bridging effect of banana fibers between the cracks. Therefore, the presence of banana fibers helps in resisting the production and progression of micro cracks in the concrete. The addition and mixing of fibers were performed following the procedure outlined in section 3.3. Therefore, the presence of banana fibers helps in resisting the production and progression of micro cracks in the concrete. The addition and mixing of fibers were performed following the procedure outlined in section 3.3. Broken surface of specimes are shown in figure 4.4. The broken surfaces of the tested specimens in Figure 4.4 reveal a uniform distribution of fibers. Some small fragments, shown in the bottom right image of Figure 4.4, are attached to the broken surface through the fibers, which indicates that structural elements containing BF can withstand spalling. This attachment is due to the bridging effect caused by the BF, resulting in enhanced durability and performance of the concrete.



Bond of fibers with surrounding matrix



Sticked small particles due to fibers



Bridging effect of fibers



Hanging of small broken particles



Pullout of banana fibers





FIGURE 4.4: Fractured Surfaces of Different Tested Specimens

4.6 Summary

The current chapter presents a comprehensive analysis of the properties of concrete mixed with different ratios of banana fibers and used petrol-engine oil (UBFRCs), compared with plain concrete (PC) and concrete mixed with only used petrolengine oil (UPC). The workability properties, densities of hard concrete, dynamic properties, mechanical properties, water absorption, mass loss, and linear shrinkage properties are discussed in detail. It is observed that the addition of BF has a positive effect on the dynamic modulus of rigidity, splitting-tensile strength, splitting-tensile toughness index, splitting-tensile pre-crack energy absorption property, splitting-tensile post-crack energy absorption property, and compressive toughness index. However, the addition of UPEO resulted in an increase in the slump value of concrete, while the addition of BF caused a reduction in the slump value of concrete. The flexural strength was reduced by the addition of UPEO, while the flexural strength was improved by the addition of BF.The water absorption property increased with increasing BF content in the concrete, while the linear shrinkage showed the opposite relationship with increasing BF content. Furthermore, the mass loss was observed to increase with increasing BF content in the concrete. The bonding between the banana fibers and surrounding matrix was found to be good when the broken surfaces of fractured specimens were examined. Overall, the results indicate that the addition of banana fibers has a positive effect on several properties of the concrete, although there are some limitations associated with the use of this material.

Chapter 5

Guidelines for Practical Implementation

5.1 Background

The paragraph discusses the results obtained from testing the impact of fiber ratio on the properties of used petrol-engine oil and banana fiber reinforced concrete (UBFRC). The study generated quantitative results graphs of stress-strain, load-deflection, and load-deformation, which demonstrated how fibres affected the mechanical properties and dynamic properties of UBFRC. The study revealed that increasing the fiber content in UBFRC resulted in improved mechanical properties, including an increase in compressive strength and modulus of elasticity. The addition of fibers also resulted in enhanced dynamic properties, such as impact resistance and energy absorption. The study demonstrated that a higher fiber content is not always better and that there is an optimal fiber ratio that can provide the best balance between mechanical and dynamic properties. The chapter also includes discussions on the practical implementation and recommendation of UBFRC in real-life applications. Overall, the study provides valuable insights into the properties of UBFRC and the impact of fiber content on its mechanical and dynamic properties. The empirical relation developed in the study can be used to optimize the use of UBFRC in different construction applications.

5.2 Optimization of Banana Fibers with Waste Petrol-Engine Oil

Table 5.1 provides details of the maximum and minimum values obtained from mechanical and dynamic tests. This information can help to select a particular combination for a specific application.

UPEO showed better properties. The addition of BF in concrete significantly influenced its properties. The effects of different proportions of BF on UBFRC properties can be seen in Figure 5.1. Some properties are significantly improved, such as the compression toughness index (CTI), while others have an adverse effect on the banana fibers, such as a decrease in compressive strength (CS). The splittingtensile properties of UBFRCs are highly improved compared to compression and flexural properties, and the flexural properties are affected in a progressive manner with the addition of fibers.Banana fibers UBFRC3+5, UBFRC3+7, UBFRC5+7, and UBFRC 3+5+7 are used and used engine oil locally available material are used. Overall, the study provides insights into the use of UBFRC in construction and the effects of adding BF and UPEO on its properties. Some properties are significantly improved, such as the com- pression toughness index (CTI), while others have an adverse effect on the banana fibers, such as a decrease in compressive strength (CS). However, the addition of UPEO resulted in an increase in the slump value of concrete, while the addition of BF caused a reduction in the slump value of concrete.

This study focuses on the optimization of banana fibers as a reinforcement material in the context of ultra-high performance fiber-reinforced concrete (UBFRC), utilizing waste petroleum engine oil (UPEO) as an additive to improve its overall performance and sustainability. By investigating the effects of various ratios and combinations of BF and UPEO on the material's mechanical properties, as well as its environmental impact, this research contributes to the ongoing efforts to develop more efficient and eco-friendly construction materials. The present study examines the effects of incorporating banana fiber (BF) and used engine oil (UEO) on the properties and performance of used petrol engine oil banana fiber-reinforced

55

concrete (UBFRC). By conducting a series of tests to evaluate the mechanical strength, durability, and sustainability of the composite, as well as its workability and environmental impact, this research aims to shed light on the potential of BF and UEO as additives in UBFRC. The results of this study could have significant implications for the development of more sustainable and cost-effective construction materials. The utilization of waste products such as used engine oil (UEO) in the production of building materials has gained increasing attention in recent years due to its potential to reduce environmental pollution and waste. overall performance and properties. By investigating the various parameters and characteristics of the resulting composite, including its water absorption, mass loss, linear shrinkage, slump value, and mechanical strength, this research contributes to a deeper understanding of the potential benefits and limitations of UBFRC, Some properties are significantly improved, such as the com- pression toughness index (CTI), while others have an adverse effect on the banana fibers, such as a decrease in compressive strength (CS). It was observed that the inclusion of used petrol engine oil (UPEO) in the concrete mixture led to a marked increase in its slump value, indicating a greater workability and ease of handling during the casting process. On the other hand, the addition of banana fibers (BF) to the composite had the opposite effect, resulting in a decrease in the slump value of the concrete, which may be attributed to the higher surface area and roughness of the fibers, leading to increased friction and resistance within the mixture.

This research project seeks to optimize the use of banana fibers (BF) as a reinforcement material in used petrol engine oil reinforced concrete (UBFRC), with the added benefit of incorporating waste petroleum engine oil (UPEO) as an ecofriendly additive to enhance the material's properties. By exploring the effects of different ratios and combinations of BF and UPEO on the mechanical and physical properties of UBFRC, including its compressive strength, flexural strength, and water absorption capacity, By analyzing the impact of UEO on the physical through a series of tests and measurements, this study aims to provide insights into the feasibility of using waste products in the development of sustainable and high-performance construction materials and to provide valuable insights into the potential of UBFRC as a more efficient and eco-conscious construction material.

Concrete	Compression			Splitting-Tensile			Flexural			Dynamic**			
Туре	\mathbf{CS}	MOE	CTE	CTI	\mathbf{SS}	STE	STI	\mathbf{FS}	FTE	FTI	ζ	Ed	Rd
	(MPa)	(GPa)	$(\mathrm{MJ/m^3})$	(-)	(MPa)	(Nm)	(-)	(MPa)	(Nm)	(-)	(%)	(GPa)	(GPa)
DC	23.72	27.77	0.167	1.68	1.5	9.92	1	3.22	0.78	1	1.76	21.3	26.6
PC	± 0.5	± 1.4	± 0.002	± 0.02	± 0.1	± 0.2	1	± 0.02	± 0.1		± 0.012	± 1.785	± 0.645
UPC	12.73	20.33	0.124	1.73	1	4.39	1	2.9	1.17	1	1.77	22.3	26.3
	± 0.2	± 0.7	± 0.006	± 0.05	± 0.2	± 0.1	1	± 0.03	± 0.02	2	± 0.051	± 1.151	± 0.412
UBFRC* with	6.8	14.9	0.141	3.43	0.96	15.4	2.23	2.64	1.46	1.1	2.83	21.6	24.3
minimum values	± 0.9	± 1.2	± 0.01	± 0.02	± 0.1	± 1.2	± 0.2	± 0.1	± 0.1	± 0.1	± 0.077	± 0.486	± 0.052
	5 + 7	5 + 7	5 + 7	5 + 7	3 + 5	3 + 5	3+5+7	3 + 5	3+5	5 + 7	3 + 7	3+5	3 + 5 + 7
UBFRC*	12.09	19.81	0.235	6.25	1.66	30.5	2.65	4.21	3.31	1.28	3.35	22.5	27.5
maximum values	± 0.2	±1.2	± 0.02	± 0.01	± 0.2	± 3.2	± 0.2	± 0.2	± 0.2	2 ± 0.1	± 0.082	± 0.912	± 0.002
	3+7	3 + 7	3 + 5	3+5+7	3+5+7	3+5+7	5+7	5+7	5 + 7	3 + 5	3+5	5+7	3+5

TABLE 5.1: Optimization of Banana Fiber Content in UBFRC

56
5.3 Application of this Research in Real Life

Concrete is used in various civil engineering applications and is subjected to different types of loadings, including mechanical and dynamic loading. The efficiency and performance of concrete are affected by these loadings, including properties like compressive strength, tensile strength, and flexural strength. Concrete durability also depends on these loadings, as cracks can be produced due to high water absorption, more linear shrinkage, and less strength of concrete in tension. Flexural strength property of concrete control rigid pavements cracking. Spalling of concrete is another issue that reduces its durability, and it can be caused by exposure to high temperatures. However, tensile strength improve spalling of concrete with the addition of fibers in the composite. Impact loading, such as blasting or vehicle collisions with concrete bridge piers, can also lead to the failure of structures. Therefore, concrete resistance against it impact loading can be improved by enhancing its dynamic modulus of rigidity and energy absorption property. Therefore, UBFRC(5+7) is more suitable as it has shown better mod- ulus of rupture and modulus of elasticity, which are key factors in the stability and durability of the rigid pavements. For compression members that go purely under uniaxial loading, UBFRC(3+7) is suitable as it has shown higher compression strength than any other type of UBFRCs.

This study explores the behavior of different types of concrete, including plain concrete (PC), Used petrol engine oil plain Concrete (UPC), and UBFRCs, by using used petrol-engine oil (UPEO) and different contents of banana fibers (BF). The specimens of UBFRC(3+7) have shown better performance against compressive loading, while the properties of UBFRC(3+5+7) are improved against tension loading and can be used for members like beams and slabs. Rigid pavements are designed by keeping in mind the flexural strength and modulus of elasticity of the concrete. Therefore, UBFRC(5+7) is more suitable for rigid pavements (light vehicles) as it has shown better modulus of rupture and modulus of elasticity, which are key factors in the stability and durability of the rigid pavements. For compression members that go purely under uniaxial loading, UBFRC(3+7) is suitable as it has shown higher compression strength than any other type of UBFRCs. UBFRC(3+7) has shown good compres

-sion strength along with some resistance against moment forces, There has been a compromise on CS with gains in CTI, SS, STI, FS and FTI. Accordingly, where compromise on CS can be made (i.e. for non-important structure like shed / storage structure), these materials (i.e. UPC BFRC) can be utilized. Therefore, UBFRC(5+7) is more suitable as it has shown better mod- ulus of rupture and modulus of elasticity, which are key factors in the stability and durability of the rigid pavements. For compression members that go purely under uniaxial loading, UBFRC(3+7) is suitable as it has shown higher compres- sion strength than any other type of UBFRCs In summary, this study provides insights into the behavior of different types of concrete and the effects of adding waste petrol-engine oil and banana fibers. It also offers recommendations for the practical implementation of UBFRC in various civil engineering applications based on its properties under different types of loadings.

5.4 Summary

In this study, optimization of banana fibers and used petrol-engine oil in UBFRCs is carried out, and recommendations are made based on strength and toughness index. The optimization is done by varying the content of banana fibers in UBFRCs and testing them under different types of loading such as mechanical and dynamic loading. Based on the results obtained, specific recommendations are made for using UBFRCs in different civil engineering applications. The developed empirical relations are used to calculate the strength properties of UBFRCs, which are then compared with the experimental properties. Overall, the study provides valuable insights into the performance of UBFRCs under different types of loading and their suitability for specific applications

Chapter 6

Conclusions and Recommendations

6.1 Conclusions

The value of waste materials is increasing with the passage of time. Some of these wastes are severe damaging to the environment. One of these hazardous wastes is used petrol-engine oil. There is a need to dump these waste in an effectice way so that the impact of environmetnal pollution causing phenomenons can be reduced. In this study, the waste materials are utilized to explores the behavior of the waste/used petrol-engine oil and banana fiber reinforced concrete (UBFRC). Various values of banana fibers (BF) have been used along with waste petrol engine oil (UPEO) to investigate the dynamic properties, mechanical properties, water absorption property, linear shrinkage property, and mass loss. Concrete is admixed with BF and UPEO in various ratios by adding cement mass. From this research, following conclusions have been made:

- UBFRC having smaller fiber lengths has highest damping ratio compared to UBFRC having longer fiber lengths in both cylinders and beamlets.
- WA, LS, ML are effected by the addition of BF and UPEO.

- UBFRC 5+7 has least water absorption.
- UBFRC 5+7 has shown least linear shrinkage.
- UBFRC 3+5 has least mass loss.
- The mechanical properties of concrete are influenced by the incorporation of the BF and UPEO.
 - There has been a significant compromise on compression strength with considerable gain in compressive toughness.
 - UBFRC 3+5+7 has highest SS and UBFRC 5+7 has highest STI.
 - UBFRC 5+7 has highest FS and UBFRC 3+5 has highest FTI.
- Both fibers breakage and fiber pullout are observed on the fractured surfaces.

Based on this research work, customized UBFRC could be utilized for different applications as there is no single combination which has shown reasonably favorable improvement in all studied properties.

6.2 Future Works

- Concrete having fibers, admixture and used petrol engine oil should be carefully studied in depth for robust outcomes.
- Durability of such concretes may also be explored.

Bibliography

- F. Aslani, G. Wan, and G. Muselin, "Development of high-performance selfcompacting concrete using waste recycled concrete aggregates and rubber granules," vol. 182, pp. 553-566, 2018.
- [2] H. M. Awais, "Optimization of Jute Fibre Content in Concrete Slabs having GFRP Rebars against Impact Loading," pp. 4-6, 2021.
- [3] N. Bekraoui, Z. El Qoubaa, H. Chouiyakh, M. Faqir, and E. Essadiqi, "Banana Fiber Extraction and Surface Characterization of Hybrid Banana Reinforced Composite," pp. 1-14, 2022.
- [4] Q. Nie et al., "Numerical simulation of fly ash concrete under sulfate attack," vol. 84, pp. 261-268, 2015.
- [5] C. Zhou et al., "Impact of freeze-thaw environment on concrete materials in two-lift concrete pavement," vol. 262, p. 120070, 2020.
- [6] X. Li et al., "A systematic review of waste materials in cement-based composites for construction applications," vol. 45, p. 103447, 2022.
- [7] K. Senthilkumar, I. Siva, N. Rajini, J. W. Jappes, and S. Siengchin, "Mechanical characteristics of tri-layer eco-friendly polymer composites for interior parts of aerospace application," in Sustainable composites for aerospace applications: Elsevier, pp. 35-53, 2018.

- [8] M.-A. Arsène, K. Bilba, and C. Onésippe, "Treatments for viable utilization of vegetable fibers in inorganic-based composites," in Sustainable and Nonconventional Construction Materials using Inorganic Bonded Fiber Composites: Elsevier, pp. 69-123, 2017.
- [9] R. Abro, X. Chen, K. Harijan, Z. A. Dhakan, and M. Ammar, "A comparative study of recycling of used engine oil using extraction by composite solvent, single solvent, and acid treatment methods," pp. 15-22, 2013.
- [10] S. Ahmed and M. Ali, "Use of agriculture waste as short discrete fibers and glass-fiber-reinforced-polymer rebars in concrete walls for enhancing impact resistance," vol. 268, p. 122211, 2020.
- [11] M. U. Farooqi, M. Ali, and B. Materials, "Contribution of plant fibers in improving the behavior and capacity of reinforced concrete for structural applications," vol. 182, pp. 94-107, 2018.
- [12] A. Ghazy, M. T. Bassuoni, E. Maguire, and M. Loan, "Properties of fiberreinforced mortars incorporating nano-silica," vol. 4, no. 1, p. 6, 2016.
- [13] A. Abushanab, W. Alnahhal, and M. Farraj, "Structural performance and moment redistribution of basalt FRC continuous beams reinforced with basalt FRP bars," vol. 240, p. 112390, 2021.
- [14] K. M. M. Rao and K. Rao, "Extraction and tensile properties of natural fibers: Vakka, date and bamboo," vol. 77, no. 3, pp. 288-295, 2007.
- [15] L. A. Pothan and S. Thomas, "Effect of hybridization and chemical modification on the water-absorption behavior of banana fiber-reinforced polyester composites," vol. 91, no. 6, pp. 3856-3865, 2004.
- [16] T. Barroqueiro, P. R. da Silva, and J. De Brito, "High-performance selfcompacting concrete with recycled aggregates from the precast industry: durability assessment," vol. 10, no. 6, p. 113, 2020.

- [17] M. Abed, R. J Nemes, and C. Materials, "Fresh properties of the self-compacting high-performance concrete using recycled concrete aggregate," vol. 71, no. 1, pp. 12-18, 2019.
- [18] C. Sun, Q. Chen, J. Xiao, W. Liu, Conservation, and Recycling, "Utilization of waste concrete recycling materials in self-compacting concrete," vol. 161, p. 104930, 2020.
- [19] M. Marvila, P. de Matos, E. Rodríguez, S. N. Monteiro, and A. de Azevedo, "Recycled aggregate: a viable solution for sustainable concrete production," vol. 15, no. 15, p. 5276, 2022.
- [20] R. K. Goyal, V. Agarwal, R. Gupta, K. Rathore, and P. Somani, "Optimum utilization of ceramic tile waste for enhancing concrete properties," vol. 49, pp. 1769-1775, 2022.
- [21] A. Alsaif and B. Materials, "Utilization of ceramic waste as partially cement substitute–A review," vol. 300, p. 124009, 2021.
- [22] G. Ma, B. Zhou, M. Zhang, J. Sanjayan, and B. Materials, "Understanding and eliminating of expansion caused by recycled glass fiber reinforced plastic powder in concrete," vol. 347, p. 128542, 2022.
- [23] M. Al-Tayeb, Y. Aisheh, S. M. Qaidi, and M. Tayeh, "Experimental and simulation study on the impact resistance of concrete to replace high amounts of fine aggregate with plastic waste," vol. 17, p. e01324, 2022.
- [24] H. Chen, L. Chow, and D. Laua, "Developing green and sustainable concrete in integrating with different urban wastes," p. 133057, 2022.
- [25] B. S. Hamad, A. A. Rteil, M. El-Fadel, and B. materials, "Effect of used engine oil on properties of fresh and hardened concrete," vol. 17, no. 5, pp. 311-318, 2003.
- [26] B. S. Hamad, A. Rteil, and B. Materials, "Effect of used engine oil on structural behavior of reinforced concrete elements," vol. 17, no. 3, pp. 203-211, 2003.

- [27] W. O. Akintunde, O. A. Olugbenga, and O. Olufemi, "Some adverse effects of used engine oil (common waste pollutant) on reproduction of male sprague dawley rats," vol. 3, no. 1, p. 46, 2015.
- [28] L. A. Thompson and W. Darwish, "Environmental chemical contaminants in food: review of a global problem," pp. 22-24, 2019.
- [29] Y. L. Yaphary, R. H. Lam, D. Lau, and B. Materials, "Reduction in cement content of normal strength concrete with used engine oil (UEO) as chemical admixture," vol. 261, p. 119967, 2020.
- [30] A. A. Mamun, H. Al-Abdul Wahhab, M. Dalhat, and Engineering, "Comparative evaluation of waste cooking oil and waste engine oil rejuvenated asphalt concrete mixtures," vol. 45, no. 10, pp. 7987-7997, 2020.
- [31] H. Taherkhani, F. Noorian, and P. Design, "Comparing the effects of waste engine and cooking oil on the properties of asphalt concrete containing reclaimed asphalt pavement (RAP)," vol. 21, no. 5, pp. 1238-1257, 2020.
- [32] A. Woszuk, M. Wróbel, and W. Franus, "Influence of waste engine oil addition on the properties of zeolite-foamed asphalt," vol. 12, no. 14, p. 2265, 2019.
- [33] J. Cai et al., "Investigation on high-viscosity asphalt binder for permeable asphalt concrete with waste materials," vol. 228, pp. 40-51, 2019.
- [34] N. Asim et al., "Wastes from the petroleum industries as sustainable resource materials in construction sectors: Opportunities, limitations, and directions," vol. 284, p. 125459, 2021.
- [35] C. Burbano-Garcia, A. Hurtado, Y. F. Silva, S. Delvasto, G. Araya-Letelier, and B. Materials, "Utilization of waste engine oil for expanded clay aggregate production and assessment of its influence on lightweight concrete properties," vol. 273, p. 121677, 2021.
- [36] B. Hasan, "Properties of Concrete Having Used Petrol-Engine Oil and Assorted Proportions of Banana Fibers," CAPITAL UNIVERSITY, pp. 2-28, 2021.

- [37] A. Rehman and M. Adnan, "AN OVERVIEW ON COMPRESSIVE BE-HAVIOUR OF COCONUT FIBER REINFORCED PLASTER," CAPITAL UNIVERSITY, pp. 10-15, 2021.
- [38] M. Ali, "Use of coconut fibre reinforced concrete and coconut-fibre ropes for seismic-resistant construction," vol. 66, no. 321, pp. e073-e073, 2016.
- [39] R. Alyousef et al., "Synergistic effects of modified sheep wool fibers on impact resistance and strength properties of concrete composites," vol. 336, p. 127550, 2022.
- [40] M. Ali and N. Chouw, "Coir fibre and rope reinforced concrete beams under dynamic loading," in Annual Australian Earthquake Engineering Society Conference, "Newcastle Earthquake-20 years on, pp. 18-20, 2009.
- [41] J. Mo et al., "Mechanical properties and damping capacity of polypropylene fiber reinforced concrete modified by rubber powder," vol. 242, p. 118111, 2020.
- [42] A. Elbehiry, O. Elnawawy, M. Kassem, A. Zaher, N. Uddin, and M. J. C. S.i. C. M. Mostafa, "Performance of concrete beams reinforced using banana fiber bars," vol. 13, p. e00361, 2020.
- [43] N. Venkateshwaran, A. Elayaperumal, and Composites, "Banana fiber reinforced polymer composites-a review," vol. 29, no. 15, pp. 2387-2396, 2010.
- [44] M. Asyraf et al., "Mechanical properties of oil palm fibre-reinforced polymer composites: A review," pp. 4-6, 2021.
- [45] S. Sbahieh, F. Tahir, and S. Al-Ghamdi, "Environmental and mechanical performance of different fiber reinforced polymers in beams," pp. 18-24, 2022.
- [46] N. Ponnambalam, S. Thangavel, G. Murali, and N. Vatin, "Impact Strength of Preplaced Aggregate Concrete Comprising Glass Fibre Mesh and Steel Fibres: Experiments and Modeling," vol. 15, no. 15, p. 5259, 2022.

- [47] F. Koksal, M. S. Yıldırım, A. Benli, and O. J. Gencel, "Hybrid effect of micro-steel and basalt fibers on physico-mechanical properties and durability of mortars with silica fume," vol. 15, p. e00649, 2021.
- [48] H. Song, J. Liu, K. He, and W. Ahmad, "A comprehensive overview of jute fiber reinforced cementitious composites," vol. 15, p. e00724, 2021.
- [49] R. Alyousef, H. Mohammadhosseini, M. M. Tahir, and H. Alabduljabbar, "Green concrete composites production comprising metalized plastic waste fibers and palm oil fuel ash," vol. 39, pp. 911-916, 2021.
- [50] A. C215-19, "Standard Test Method for Fundamental Transverse, Longitudinal, and Torsional Resonant Frequencies of Concrete Specimens," ed: ASTM International West Conshohocken, PA, USA, pp. 1-5, 2019.
- [51] M. Khan, A. Rehman, M. J. C. Ali, and B. Materials, "Efficiency of silicafume content in plain and natural fiber reinforced concrete for concrete road," vol. 244, p. 118382, 2020.
- [52] A. Zia, M. Ali, and B. Materials, "Behavior of fiber reinforced concrete for controlling the rate of cracking in canal-lining," vol. 155, pp. 726-739, 2017.
- [53] Yaphary, Y.L., Lam, R.H. and Lau, D., 2020. Reduction in cement content of normal strength concrete with used engine oil (UEO) as chemical admixture. Construction and Building Materials, 261, p.119967.
- [54] G. E. Abdelaziz, Utilization of used-engine oil in concrete as a chemical admixture. Benha University, Egypt" pp. 14-17, 2011.
- [55] Hamad, B. S., Rteil, A. A. (2003). Effect of used engine oil on structural behavior of reinforced concrete elements. Construction and Building Materials, 17(3), pp. 203-211.
- [56] Shafiq, N., Nuruddin, M. F., Beddu, S. Properties of concrete containing used engine oil. International Journal of Sustainable Construction Engineering and Technology, 2(1). pp. 18-21, 2011.

[57] Wu, F., Yu, Q., Liu, C., Brouwers, H.J.H., Wang, L. and Liu, D., 2020. Effect of fibre type and content on performance of bio-based concrete containing heat-treated apricot shell. Materials and Structures, 53, pp.1-16.